

The Role of Trust in Online Lending

Christoph Bertsch, Isaiah Hull, Yingjie Qi and Xin Zhang

November 2017

WORKING PAPERS ARE OBTAINABLE FROM

www.riksbank.se/en/research

Sveriges Riksbank • SE-103 37 Stockholm Fax international: +46 8 21 05 31 Telephone international: +46 8 787 00 00

The Working Paper series presents reports on matters in the sphere of activities of the Riksbank that are considered to be of interest to a wider public. The papers are to be regarded as reports on ongoing studies and the authors will be pleased to receive comments.

The opinions expressed in this article are the sole responsibility of the author(s) and should not be interpreted as reflecting the views of Sveriges Riksbank.

The Role of Trust in Online Lending^{*}

Christoph Bertsch

Sveriges Riksbank

Isaiah Hull

Sveriges Riksbank

Yingjie Qi Stockholm School of Economics Xin Zhang Sveriges Riksbank

Sveriges Riksbank Working Paper Series No. 346 November 2017

Abstract

We study the impact of trust on the expansion of online lending in the U.S. over the 2008-2016 period. Using data from the largest platform, we demonstrate that a misconduct-driven decline of trust in traditional banking is associated with a statistically and economically significant increase in online lending at the state level. To the contrary, increased social trust strengthens in-person, bank-based borrowing and informal borrowing, reducing the demand for impersonal online lending. Both of these effects operate primarily through borrowers. We also use a shock that affects only investors to demonstrate that distrust in traditional finance increases participation in online lending.

Keywords: financial development, consumer loans, bank misconduct, FinTech **JEL Classification:** A13, G00, G21, K00

^{*}Christoph Bertsch, Isaiah Hull and Xin Zhang are at the Research Division of Sveriges Riksbank, 103 37 Stockholm, Sweden. Yingjie Qi is at the Department of Finance of the Stockholm School of Economics, 111 60 Stockholm, Sweden. We would like to thank Bo Becker, Giancarlo Corsetti, Julian Franks, Mariassunta Gianetti and Per Strömberg for comments, as well as seminar participants at the Graduate Institute Geneva and Sveriges Riksbank. All remaining errors are ours. The opinions expressed in this article are the sole responsibility of the authors and should not be interpreted as reflecting the views of Sveriges Riksbank.

1 Introduction

Trustworthiness matters in financial transactions.¹ Hence, the expansion of new financial services, such online lending, may be constrained by the level of trust among prospective market participants. Similarly, a decline in faith in traditional forms of finance could generate growth in online lending by driving borrowers away from banks. Such considerations are especially relevant in the aftermath of the Great Recession, where widespread misconduct in the financial sector gave rise to the concern that "fraud has become a feature and not a bug" (Zingales (2015), p.19). By eliminating the middleman, online lending platforms are able to reduce investor apprehension about opportunistic behavior that is often associated with traditional financial intermediaries. From a borrower's perspective, online lending is relatively impersonal, which suggests that increased interpersonal trust will primarily benefit bank-based borrowing, where personal interactions with bank employees remain common.

Using nearly-complete loan and application histories from the two largest and oldest U.S. platforms, LendingClub.com and Prosper.com, we measure the impact of time and geographic variation in trust on state-level online lending. Our measures of trust are divided into three groups: 1) interpersonal trust; 2) trust in traditional banking; and 3) trust in traditional finance. Following the literature (e.g., Gambetta (2000); Guiso et al. (2008)), we define a borrower's interpersonal trust as the subjective probability she assigns to the chance of being cheated by others, and use survey responses from the General Social Survey (GSS) to measure it. Similarly, borrower trust in the traditional banking sector is defined as an individual's subjective probability of being cheated by banks. We construct a proxy for trust in banks using financial misconduct data from the Consumer Financial Protection Bureau (CFPB). Finally, we use an analogous definition for investor trust in the traditional financial sector more generally, and measure this using the geographic distribution of Madoff scandal victims, as in Guiso (2010).

We first examine how borrower trust in traditional banks affects online lending at the state level. Controlling for local macroeconomic conditions, loan-borrower characteristics, and geographic and time fixed effects, we show that lower trust in traditional banking is associated with a higher level of online lending. In particular, an increase of one complaint

¹E.g. see Guiso et al. (2004, 2008, 2013), Giannetti and Wang (2016), and Rau (2017).

per bank branch is associated with a 6% increase in the ratio of online debt to total debt² for the median state. This finding also appears to be borrower-related, which indicates that bank misconduct may drive existing customers away from traditional banking to online platforms.

In contrast to our results for trust in traditional banks, we find that higher interpersonal trust weakens online lending growth, since online lending is perceived to be impersonal relative to bank-based borrowing. In particular, a one standard deviation increase in personal trust is associated with a -11% decline in the online debt to total debt ratio for the median state. We also show that interpersonal trust operates in the same direction on credit card debt, which is also impersonal, but in the opposite direction on mortgage debt, which often requires interaction with bank staff. Furthermore, this result holds when our measure of trust in traditional banking is included as a control, which suggests that the two measures have distinct effects on borrowing. It also holds in both application and loan data.

In addition to these findings, we also identify which borrower segments are most sensitive to variation in trust levels. The 2015 FDIC National Survey of Unbanked and Underbanked Households shows that a low level of trust in banks is a significant impediment for underserved borrowers seeking financial services. Macro-level evidence points to a positive association between unemployment and distrust in institutions-and, in particular, banks (Stevenson and Wolfers 2011)-which suggests that a low level of trust in financial institutions is likely to be particularly important for underbanked households who often lack stable employment and have low credit ratings. In fact, we find that a lower level of trust in banks and a lower level of interpersonal trust are both positively associated with a higher share of online lending in the low credit rating segment.

Finally, turning to the supply side, we test whether a decline of investor trust in traditional finance increases online lending. We use the locations of the Madoff scandal victims to capture geographic variation in this category of trust. Since the scandal largely pre-dates our sample, there is no time variation to exploit. We find that investment in online lending platforms is affected positively by exposure to the Madoff scandal, which is consistent with the conjecture that it shattered the confidence of wealthy investors, shifting interest to

 $^{^{2}}$ We refer to all household debt not channeled through online lending as "total debt" or "bank debt."

alternative financial investments, such as online lending.

Our analysis complements the extensive literature on trust by exploring its role in the financial development of online lending. Generalized trust-that is, interpersonal trust and trust in institutions-has been shown to be positively associated with financial development (e.g., Guiso et al. (2004)) and appears to be an important determinant of the crowdfunding boom at the international level, as shown in a key contribution by Rau (2017). Our paper is the first to study how different dimensions of trust affect online online lending. It also advances the literature by expanding our understanding of the scope of financial disintermediation that has occurred under FinTech.

The rest of the paper proceeds as follows. Section 2 reviews the related literature. Section 3 presents the theoretical framework and develops hypotheses. Section 4 provides background information on the online lending market and describes the data. The empirical results are presented in section 5. Finally, section 6 concludes. All tables are located in the Appendix.

2 Related literature

Our paper relates to three distinct strands of literature. First, it builds on the existing empirical research on social trust, connecting to the literatures on both interpersonal trust and trust in institutions. Second, our paper contributes to the literature on financial development and growth with a focus on financial innovation. And third, we complement the emerging literature on consumer credit and online lending that studies the micro- and macro-determinants of investor financing, as well as borrower behavior.

Trust. The literature proposes different definitions for the concept of trust. We use the one proposed in Gambetta (2000), which is tightly connected to beliefs.³ For online lending, Duarte et al. (2012) demonstrate that having a trustworthy appearance matters, and that it affects both investor and borrower behavior. This was shown using data from Prosper.com, a U.S.-based online lending platform, for the years in which investors were able to access borrowers' pictures. Several papers also document trust as an important determinant of

³See Fehr (2009) for a review of the "Economics and Biology of Trust."

stock market participation.⁴ For credit markets, trust appears to be even more important, since debt claims are characterized by a limited upside for investors. Consequently, investors have more to fear from being cheated by borrowers. However, trust in lenders and the perception of fair treatment also play an important role for borrowers. Guiso et al. (2013) find that lower trust in banks makes it more likely that borrowers strategically default on their mortgage debt. Moreover, there is an extensive literature documenting the response to perceived unfair treatment in various areas (e.g., Xia et al. (2004)), suggesting that it also influences borrowers' willingness to switch to online lending platforms.

Financial innovation. Guiso et al. (2004) document the important role played by trust and social capital, more generally, in financial development. Financial development, in turn, has been demonstrated to be a facilitator of economic growth, as argued, e.g. by Rajan and Zingales (1998). Rau (2017) suggests that the international expansion of crowdfunding is positively related to interpersonal trust, as well as the quality of regulation, the degree of financial development, and internet access.

In light of the existing literature, the deterioration of trust in the traditional banking industry after the Great Recession (e.g., Corsetti et al. (2010)) may have lowered the barriers to entry for new players from the FinTech sphere, fueling disintermediation in some market segments targeted by online lending platforms (FSB 2017). Since the widespread financial misconduct before and after the crisis involves many of the largest banks (Sakalauskait 2016), our measure of distrust in banks is based on misconduct, which should be particularly relevant for the sample period under consideration.

Online lending. Crowdfunding platforms have enjoyed rapid growth in recent years and have received increased attention in the literature.⁵ We focus on online lending, which is the dominant worldwide form of crowdfunding (Rau 2017).⁶ Within online lending, consumer credit is the largest market segment and tends to attract high-risk borrowers who want small

 $^{^{4}}$ See Guiso et al. (2008) for the effect of a lack of generalized trust and interpersonal trust, and Giannetti and Wang (2016) for the effect of a deterioration of trust due to corporate fraud.

 $^{{}^{5}}$ For a recent review of the literature on crowdfunding, see Belleflamme et al. (2015). Specific to crowdlending (or online lending), see Morse (2015).

⁶For an in-depth overview see the Cambridge Center for Alternative Finance (CCAF) Benchmarking Reports referenced in Rau (2017).

loans, a slice of the market that is underserved by traditional banks (De Roure et al. 2016). Many borrowers in this segment also use online platforms to increase their total borrowing capacity (Demyanyk et al. 2017). Online consumer credit is often uncollateralized and can be seen as a substitute for credit cards and other forms of consumer credit. Lending platforms operate with significantly lower costs than traditional banks and specialize in automated credit scoring. This can give FinTech lenders an advantage in screening high-risk borrowers (Einav et al. 2013), which may allow them to extend more generous loans to medium-risk borrowers. The expansion of online lending appears to be highest in regions where traditional banks are absent or capital constrained.⁷

A number of papers employ the peer-to-peer (P2P) online lending market as a laboratory to study different micro aspects of lending, such as the role of informational frictions, using U.S. data from the Prosper.com⁸ and LendingClub.com⁹ consumer credit platforms. However, macroeconomic developments have also been shown to play an important role.¹⁰ From an investor's perspective, Lin and Viswanathan (2016) document substantial home bias in online lending, which resonates with the extensive home bias literature in economics and finance. The implication for our paper is that shifts in investor attitudes in a region are likely to have measurable implications for regional online lending volumes.

3 Theoretical framework and hypothesis development

We reviewed an extensive literature that documents the historical role of trust in financial markets. With the emergence of online lending, different varieties of trust have surfaced

⁷See, e.g., Havrylchyk et al. (2017).

⁸Papers using data from Prosper.com study the role of soft information, such as the appearance of borrowers (Duarte et al. 2012; Pope and Sydnor 2011; Ravina 2012; Gonzales and Komarova Loureiro 2014), the importance of screening in lending decisions (Iyer et al. 2015; Hildebrand et al. 2016; Balyuk 2016), the herding of lenders (Zhang and Liu 2012), the importance of geography-based informational frictions (Lin and Viswanathan 2016; Senney 2016), the auction pricing mechanism that existed prior to 2011 (Chen et al. 2014; Wei and Lin 2015), and the ability of marginal borrowers to substitute between financing sources (Butler et al. 2015).

⁹There are papers using data from LendingClub.com to study adverse selection (Hertzberg et al. 2015) and retail investor risk-aversion (Paravisini et al. 2016).

¹⁰Crowe and Ramcharan (2013) study the effect of home prices on borrowing conditions. Bertsch et al. (2017) study monetary normalization and online lending.

as links in the chain that connect FinTech to traditional banking. The dominant forms of consumer credit, traditional bank loans and credit card debt, have faced growing competition from online lending platforms, especially in the segment for uncollateralized, high-risk credit debt for underbanked households.

3.1 Theoretical framework

From a conceptual viewpoint, we propose a stylized theoretical model that emphasizes the borrower perspective. This allows us to gain insights into how different dimensions of trust could affect the development of the online lending market. We allow borrowers in the model to choose between different sources of funding and assign them a payoff that depends negatively on the borrowing costs and on the probability of being cheated or treated unfairly. This might entail unfair contractual terms, opaque or unpredictable fees, or deliberate missales. We assume that personal interaction can help borrowers to build trust with lenders, and lenders to build trust with borrowers. The empirical literature suggests that this is especially true for repeated interactions in borrower-bank relationships. In a lending context, it can help to convey soft information during an interaction with a loan officer. This personal, relationship-specific component of lending favors traditional banks.¹¹

On the other hand, online lending platforms have a clear cost advantage over traditional banks: credit scoring is automated and there is no need to maintain a branch network. We assume that online platforms can leverage FinTech to offer cheaper loans based on hard information to the borrower segments they target, as long as there is no important soft information or relationship component that can be captured by traditional banks.

Formally, we consider a simultaneous-move, one-shot game with a continuum of borrowers of mass m > 0, indexed by *i* with interest in obtaining a long-term, uncollateralized loan of fixed size, L = 1. The credit market is competitive and uncollateralized. On the supply side, there is a large number of traditional banks and online lending platforms that compete

 $^{^{11}}$ We acknowledge that an appearance-based trust measure is likely to be positively associated with the chances of individual borrowers to obtain a loan from a bank or from an online lending platform, where borrowers were able to post pictures in the initial years (see Duarte et al. (2012)). The survey-based interpersonal trust measure elicits, however, a different dimension of trust that is independent of person-specific traits.

by setting lending rates. For simplicity, all agents are risk-neutral. The cost of traditional banks to issue one loan is f_B and the cost of online lenders to issue one loan is $f_O \in (1, f_B)$.

We aim to capture the heterogeneous ability of borrowers to take advantage of an inperson relationship with a traditional bank that allows them to convey soft information and to improve monitoring via repeated interaction.¹² To do this in a stylized way, we assume that borrowers have a baseline repayment probability of 0 and face an idiosyncratic $cost <math>c_i \in U[0, \overline{c}]$ to further increase the repayment probability to q if they decide to incur the cost and take out an in-person loan from a traditional bank, where $\overline{c} > 0$ and p < q < 1.

Let \bar{r} denote borrowers' reservation interest rate. We assume that borrowers have a lack of trust in both types of lenders, since they believe they may be cheated or treated unfairly with a certain probability. For simplicity, the expected utility cost of being cheated by banks is given by $\tau_B > 0$, while the expected cost of being cheated by online lenders is $\tau_O > 0$.

The timing is as follows. First, borrowers decide whether to incur a cost that allows them to increase their repayment probability with traditional banks. Second, lenders simultaneously offer rates to borrowers who then decide which offer to accept. In effect, banks compete by offering one rate to borrowers who are known to have an increased repayment probability when taking an in-person laon and a different rate to all remaining borrowers, while online lending platforms compete by offering a single interest rate to all borrowers.

Result 1 below describes the credit market equilibrium and summarizes the key comparative statics that guide our development of hypotheses.

Result 1 A sufficient condition for the existence of a credit market equilibrium with participation of all borrowers is given by:

$$\min\{f_B/q + \tau_B + \overline{c}, f_B/p + \tau_B, f_O/p + \tau_O\} \le \overline{r}.$$
(1)

All loans are issued by traditional banks if the relative trust in banks is high:

$$\min\{f_B/q + \bar{c}, f_B/p\} < f_O/p + (\tau_O - \tau_B),$$
(2)

¹²We emphasize the benefits from relationship banking and repeated interaction, which we capture in a reduced from. We acknowledge the extensive literature on signaling. Important screening tools that have been discussed include collateral and loan quantities (Bester 1985).

which is the case if $(\tau_O - \tau_B)$ is high. If inequality (2) is violated and $f_B/q < f_O/p + (\tau_O - \tau_B)$, then a fraction \hat{c}/\bar{c} of borrowers incur the cost for the relationship investment and accept a loan from traditional banks, while all others do not incur the cost for the relationship investment and accept a loan from online lending platforms, where:

$$\hat{c} \equiv f_O/p - f_B/q + (\tau_O - \tau_B). \tag{3}$$

The total online lending volume, $m(1 - \hat{c}/\bar{c})$, is increasing in τ_B , f_B and \bar{c} , while it is decreasing in τ_O and f_O . Finally, if inequality (2) is violated and $f_B/q \ge f_O/p + (\tau_O - \tau_B)$, then all borrowers do not incur the cost and exclusively accept loans from online platforms.

3.2 Hypothesis development

Guided by the literature and the theoretical framework above, we design empirical tests to measure the relationship between interpersonal trust, trust in traditional banking, trust in traditional finance, and the expansion of online lending. We first introduce the hypothesis tests and, thereafter, present the corresponding econometric models.

Hypothesis 1: trust in traditional banking (borrowers)

H0: Deterioration of trust in traditional banking, caused by bank misconduct, is positively correlated with the expansion of online lending at the state level in the US.

One implication of Result 1 is that the entry and expansion of online lending platforms in the consumer credit market should depend on the level of trust in banks. Result 1 predicts that the deterioration of trust in traditional banks after the financial crisis, captured by an increase in τ_B relative to τ_O , reduces the barrier to entry for online lending platforms and is associated with an increase in the market share of online lending.

We relate the measure of state-level online lending development to CFPB-reported bank misconduct using the following empirical model:

$$Y_{i,t} = \beta_1 \text{CFPB complaints}_{i,t} + \gamma X_{i,t} + (A_i + B_t) + \epsilon_{i,t}.$$
(4)

The dependent variable is the ratio of P2P debt (millions) to total debt (tens of billions) in state *i* in month t.¹³ The bank misconduct variable is computed as the number of consumer complaints filed to the CFPB in the same month. Our specification also includes state fixed effects, A_i , and time fixed effects, B_t . We estimate the model with OLS.

Since online lending platforms operate with significantly lower costs than traditional banks and specialize in automated credit scoring, FinTech lenders have an inherent advantage in screening high-risk borrowers (Einav et al. 2013). Thus, we might expect that the deterioration of trust in banks has a disproportionately large impact on the high-risk borrower segment. In fact, Result 1 suggests that a cost advantage for platforms screening high-risk borrowers, i.e. an increase in $f_B - f_O$, is associated with a larger share of online lending in this segment.

Hypothesis 2: interpersonal trust (borrowers)

H0: States with higher interpersonal trust should experience less credit demand in the online lending market, since online platforms are perceived to be impersonal relative to traditional banking and informal lending from friends and family.

Another implication of our theoretical framework is that borrowers will find loans from banks more attractive if it is less costly for them to take advantage of in-person relationships with traditional banks. Result 1 predicts that a lower cost-to-invest in a borrower-bank relationship, captured as a reduction in \bar{c} , is associated with a higher lending volume for traditional banks relative to online lending. Furthermore, higher interpersonal trust may also give borrowers better access to alternative sources of funding, such as informal credit from family and friends. Such credit is often strictly preferred by borrowers and easier to obtain. In the context of our model, it amounts to a lower m, which is associated with a reduction in online lending. Either way, H0 follows.

To test our hypothesis, we regress the online lending development measure on the social trust measures, computed using the GSS survey question on interpersonal trust. The

¹³Note that we use millions for P2P debt and tens of billions for total debt to deal with scaling issues, since the P2P market is small relative to the financial sector for much of our sample. Importantly, however, we will show that the effect of trust on the P2P market itself is economically significant.

regression is specified as follows:

$$Y_{i,t} = \beta_1 \text{Interpersonal trust}_i + \gamma X_{i,t} + (B_t) + \epsilon_{i,t}.$$
(5)

We use a cross-sectional measure of social trust that is computed as the average of the GSS's social trust question over the period 1973-2006 (Algan and Cahuc 2014). This captures the slow moving component of social trust, which the literature suggests is most salient.

Hypothesis 3: trust in traditional finance (investors)

H0: Investor trust in traditional banking is negatively correlated with investor participation in online lending. Thus, negative shocks to investor trust will lead to faster growth in online lending.

We treat the impact of the Madoff investment scandal in late 2008 as a negative shock to trust in traditional finance. We argue that the discovery of large scale fraud may have caused investor attitudes to shift, increasing interest in alternative investments, such as online lending. While traditional finance was tarnished by the former Nasdaq Chairman's scandal, online lending platforms, which offer a high level of transparency and information provision, may have actually gained. Thus, we would expect the supply of funds to the online market to increase, lowering borrowers' funding costs on online lending platforms.

In the context of our theoretical framework, such a shift in relative funding conditions can be captured as a decrease in $f_O - f_B$. Hence, Result 1 predicts an increase in total online lending. The econometric model is set up as follows:

$$Y_{i,t} = \beta_1 \text{Madoff victims}_i + \gamma X_{i,t} + (B_t) + \epsilon_{i,t}.$$
(6)

The Madoff victim measure is given by the number of investors who suffered losses in the scandal at the state level, as in Guiso (2010). We include only time fixed effects in the specification because the Madoff scandal data is cross-sectional.

In addition to the state-level regressions, we also investigate the relationship between trust measures and subgroup differences in applicant credit quality. We also use loan application level regressions to measure the impact of trust in traditional banking on the extensive and intensive borrowing margins.

4 Data and descriptive statistics

Peer-to-peer (P2P) online lending first emerged in the U.S. in 2005 in the form of crowdfunding. *Prosper.com* was the first U.S.-based platform, followed by *LendingClub.com*, which has been the market leader for several years. According to a Federal Reserve Bank of Cleveland study, U.S. P2P lending grew by an average of 84% per quarter between 2007 and 2014 (Demyanyk 2014). More recently, online lenders have started to transition from crowdfunding, which entails raising small amounts of funds from multiple lenders, to a mix of crowdfunding and marketplace lending, which involves securing wholesale funding from institutional investors. The accounting firm PricewaterhouseCoopers expects online lending platforms to reach 10% of revolving US consumer debt by 2025.¹⁴

Our primary dataset consists of a panel of 1.7 million loan-borrower observations from LendingClub, which operates the largest online platform for consumer credit in the US and was founded in 2006.¹⁵ As of October of 2017, LendingClub has more than 1.5 million customers, including both investors and borrowers, and has originated loans in excess of \$28 billion. While LendingClub's base of institutional investors has grown strongly since 2014, retail investors participation remains significant. Since LendingClub only recently expanded to the small business loan and auto refinancing segment, we focus exclusively on personal uncollateralized loans. Borrowers request loans ranging from \$500 to \$40,000 with a maturity of 3 to 5 years. The median borrower has a loan size of \$13,000, an interest rate of 13%, a yearly income of \$65,000, an employment duration of 6 years, and a low proprietary credit rating. The personal loans issued by LendingClub are used for a variety of purposes, including debt consolidation, large durable good purchases, and unexpected expense financing. The highest rated borrowers may have normal access to traditional sources of credit from banks and credit cards, but the lowest rated borrowers are likely to be underserved.

¹⁴See market study by PricewaterhouseCoopers (2015).

¹⁵We use LendingClub as our primary subject of study because Prosper SEC filings are incomplete over the first four years of our sample.

After a prospective borrower submits an application, the platform collects self-reported and publicly available information, including the borrower's credit history. LendingClub uses a credit model to decide on the borrower's qualification for the loan, to assign a credit score, and to set a fixed interest rate and repayment schedule. The process is fast and qualified borrowers can expect to receive an offer within 24 hours. The platform provides a large set of loan-borrower characteristics to investors and divides the market into two distinct segments: fractional and whole. The fractional loan market is where a crowd of investors screens loans posted on the platform and funds individual borrowers in \$25 increments. The whole loan market is where individual borrowers are matched with large investors who purchase entire loans. While the former market is dominated by retail investors, the latter market is dominated by institutional investors. Individual loan applications are allocated to the fractional or whole loan market by the platform and have no influence on it. We observe whether individual loan applicants successfully obtain funding and from which market segment. Provided borrowers accept the loan, the total funding volume (net of an origination fee) is disbursed. LendingClub offloads the risk to lenders and then services the loan throughout its duration, which includes monthly installment transfers from borrowers to lenders.

LendingClub (as well as Prosper) generates fee income that is growing in transaction volume. Specifically, LendingClub's fee structure for fractional loans consists of the following: 1) an origination fee of 1-6%, paid by borrowers at loan disbursement; 2) a servicing fee of 1% on the payments transferred to lenders; and 3) a set of collection fees imposed for late payment and default. The servicing fee differs for the whole loan market.

In addition to our primary dataset for LendingClub, we also have a secondary dataset with application data for Prosper, which spans the 2012-2017 period. Prosper's platform design is very similar to LendingClub's. Our application data for both LendingClub and Prosper comes from 424B3 filings, which we retrieved from the SEC's Edgar database. Our loan data for LendingClub comes from LendingClub's loan book, retrieved from their website. The SEC filings contain all application data, including loan and borrower characteristics. The loan book contains the set of loans originated, as well as loan characteristics, borrower characteristics, and repayment status updates. See Table I for the list of variable definitions and Table II for the associated summary statistics of our primary dataset.

We also collect information on variables related to the core hypotheses we test in the

paper: 1) a survey-based measure of interpersonal trust; 2) factors that affect trust in traditional banking, such as instances of bank misconduct; 3) debt origination data from the Federal Reserve Bank of New York; 4) FDIC bank branch data; and 5) state-level economic and demographic controls. The period we study begins with the earliest available data on P2P lending in the U.S. in 2008 and continues until 2016. Data sources (3) and (4) are used primarily to normalize our variables of interest.

The survey-based measure of trust was obtained from the General Social Survey (GSS), which is conducted biennially by the National Opinion Research Center (NORC) at the University of Chicago. This contains a measure of interpersonal trust, is available for the 1973-2016 period, and includes both the region of residence and region of residence at age 16 for each respondent. We focus on interpersonal trust for the current region of residence and average over all observations in a given state-year. This yields a measure with a scale of 0 to 1, which varies between 0.18 and 0.62 in our sample.

We also collect data on several factors that affect trust, including exposure to financial misconduct. We use the following data for financial misconduct: 1) the Consumer Financial Protection Bureau's (CFPB) Consumer Complaint Database; and 2) the list of Bernie Madoff's victims. The CFPB data contains the name of the bank, the time of the complaint, and the location of the customer. We compute the total number of complaints per state. We then normalize this by the number of bank branches in the state, which we take from the Federal Deposit Insurance Corporation's (FDIC) database. The average state has 0.20 complaints per branch. We hand collected the list of Bernie Madoff's victims' identities and locations from the New York Times's website. We then computed the number of victims per state.

5 Empirical results

We test our hypotheses listed in Section 3 and present the results. We also include robustness checks to further nuance our empirical test results. All tests conducted using our primary dataset, but are robust to the inclusion of secondary data from Prosper.

5.1 Borrower trust in traditional banking

The first hypothesis claims that a lack of trust in financial institutions drives borrowers from traditional banks to online lenders. To measure trust, we use consumer complaints filed to the U.S. Consumer Financial Protection Bureau (CFPB). The underlying assumption is that borrowers in states that experience a high number of complaints per bank branch–an indication of bank misconduct or customer mistreatment–will tend to have lower levels of trust in traditional banks.

In Table III, we test Hypothesis 1 by regressing the ratio of online debt (m\$) to total debt (10b\$) on the average number of consumer complaints per bank branch at the state level. We use complaints per bank branch to measure distrust in traditional financial institutions. In total, we have 2839 state-month observations. In the even-numbered columns, we use a specification that controls for the following variables at the state-level: population density, GDP, the unemployment rate, and the population size. We also control for borrower characteristics by including the state-level averages for the interest rates charged on loans, the gross income levels of borrowers, and the number of years employed. In columns 3 and 4, we include fixed effects to control for time-invariant state level characteristics. Columns 5 and 6 include both state and year fixed effects to control for time-invariant, state-level characteristics and other common sources of time series variation that may have contributed to growth. In columns 7 and 8, we replace year fixed effects with year-month fixed effects, along with state fixed effects. This imposes more rigorous control on the time trend and the business cycle component to capture any shocks that occurred in a particular month.

Our estimate of the impact of borrower trust in traditional banking on P2P borrowing is positive and statistically significant at the 1% level. Column 8 contains our preferred regression specification. We find that increasing the number of complaints by 1 per bank branch is associated with a 0.009 increase in the online debt ratio. Stated differently, a one complaint increase per bank branch is associated with an increase in the ratio of online debt to total debt by 6% for the median state. This suggests that distrust in traditional financial institutions has an economically significant impact on the growth of online lending.

We also analyze the impact of CFPB complaints on the extensive and intensive margins using loan application-level information. We compute the impact on the extensive margin by regressing the fraction of P2P borrowers in the state's population on the average borrower pool quality variables and on state level controls. The intensive margin is measured using a regression that links the loan application size to borrower-loan characteristics. From Table IV and V, we find that the results for the CFPB are mostly driven by the extensive margin, rather than the intensive margin. Equivalently, CFPB complaints at the state level motivate higher online lending growth, but do not affect borrowers' decisions about the loan amount.

These results align well with intuition. Namely, a higher level of distrust in banks is associated with a higher number of borrowers deciding to switch to online lending. Conversely, there is no compelling reason to expect that the typical borrower who switches would request more funds. Thus, it seems plausible that state-level gains would be driven by extensive margin gains. This is also consistent with our theoretical framework, which makes predictions about the extensive margin.

5.2 Interpersonal trust

To test Hypothesis 2, we regress the fraction of P2P debt on a cross-sectional measure of trust-namely, the average of positive responses to the interpersonal trust question from the General Social Survey (GSS) over the period 1973-2006, which pre-dates our sample (See Algan and Cahuc (2014)). Previous work in the literature argues that interpersonal trust is a slow-moving variable within a given society, which suggests that regional variation is likely to be most meaningful.¹⁶ As predicted, the relationship between interpersonal trust and P2P debt is negative, which suggests that higher levels of interpersonal trust are associated with less online lending at the state-level. Both measures yield similar results and are robust.

In Table VI, the cross sectional variable from the GSS is used as the measure of interpersonal trust. Note that the cross-state variation of trust in other people is a weight between zero and one. In total, we have 4297 observations at the state-month level, although the GSS trust measure is kept constant over time. In even-numbered columns, we add a list of additional controls, including the state-level population density, GDP, the unemployment rate, and the population size. As in Table III, we control for other borrower characteristics

 $^{^{16}}$ Trust is rooted in history and social norms (Tabellini 2008). See, e.g., Algan and Cahuc (2010) and Dohmen et al. (2012) on inter-generational transmission of trust.

by including state-level averages for the interest rate charged on the loans, the gross income level of the borrowers, and the number of years employed. Since the measure of interpersonal trust is time-invariant for each state, we cannot include state fixed effects to absorb the unobserved variables that might drive the result. We include year fixed effects in columns 3 and 4, and year-month fixed effects in column 5 and 6. This captures the time-varying business cycle component and anything that occurred systematically to the economy. The coefficient estimate does not change substantially across specification, and is roughly -0.15 and statistically significant at the 1% level. This suggests that a one standard deviation increase in interpersonal trust is associated with -11% reduction in online lending's share for the median state.

At least two factors can plausibly explain our results. First, online lending provides a marketplace where lenders and borrowers do not have to interact personally. This differs from traditional banking-based borrowing, which often requires interaction between the borrower and a bank employee. In this respect, the disintermediation facilitated by online lending reduces the importance of interpersonal trust. Lending decisions are made using algorithms, rather than with hard and soft information elicited through human interaction. Second, borrowing from friends and family members constitutes an important channel for inexpensive credit, especially in the event of an unexpected expense or crisis. In states with high interpersonal trust, we expect individuals to have closer ties that enable informal borrowing through this channel. Even though online borrowing is a convenient and direct way of obtaining credit, informal lending will be preferred for more borrowers in states with higher interpersonal trust. We will examine this more formally below.

To test whether interpersonal trust is the relevant channel, we repeat the exercise from Table VI, but replace P2P's debt share with credit card borrowing (Table VII) and mortgage borrowing (Table VIII). We find a negative relationship between interpersonal trust and credit card borrowing, indicating that our findings for the online lending market also hold for credit card debt. This is what we would expect, since credit card borrowing is similar to online lending in that it is an impersonal process with no interviews required. It is also similar to online lending and informal lending in that it may be used to handle budgetary shocks (e.g. medical bills, car repairs, etc.), since it is unsecured and may be used with short notice. To the contrary, none of this is true for mortgage debt, which is secured by the value of the home, often requires a personal interaction with a bank employee, and has a larger principle amount and a longer maturity. In line with this explanation, we find a positive, rather than negative, relationship between interpersonal trust and mortgage borrowing. This supports our claim that in-person borrowing and informal lending gain from higher interpersonal trust, but online borrowing and credit card borrowing do not.

5.3 Investor trust in traditional finance

Thus far, we have exclusively focused on the demand side of P2P credit by using loan application data. To capture how investor preferences shaped platform growth, we examine the Madoff investment scandal, which arguably influenced only investor attitudes towards risky investments. In Table IX, we regress the originated loan volume that was funded by both investor types as a share of total borrowing on the number of Madoff victims in the same state, a dummy indicating whether the loan was wholesale or retail, and the interaction of both terms. In total, there are 4653 observations at the state-month level. The specifications are similar to those in Table VI. In even-numbered columns, we add a list of state-level controls, including population density, GDP, the unemployment rate, and population size. We also control for other borrower characters at the state level that may drive results, including the average interest rate charged on the loans, the average gross income level of the borrowers, and the average number of years employed. We include year fixed effects in columns 3 and 4, and year-month fixed effects in columns 5 and 6. The coefficient from the last column indicates that increasing the Madoff victim count by 1000 in a state is associated with a 0.006ppt increase in the online-to-total lending share. This is equivalent to a 4.7% increase in the ratio of online debt to total debt in the median state. While this effect is somewhat smaller relative to our findings for borrowers, it is likely to be biased downward, since it relies on investor home bias, which may be relatively weak.

5.4 Robustness checks

Above, we discussed the empirical tests for the hypotheses presented in Section 3. We found support for the hypothesis that reduced borrower trust, driven by bank misconduct, increases participation in the online lending market. We also found that social trust, measured by interpersonal trust, hinders the expansion of online lending. In the following section, we will analyze subgroups with heterogenous borrower characteristics. We will then consider a regression specification that includes both bank misconduct and interpersonal trust.

5.4.1 Borrower subgroups

To identify which borrower quality group is more sensitive to bank misconduct and other trust measures, we repeat the exercises from Table III, Table VI, and Table IX, but for borrowers with high and low credit ratings. Fourty percent of the borrowers have a rating of either "A" or "B." We consider these borrowers to be high quality and assign a value of one to them for the dummy variable "highrating." The rest of the borrowers have ratings ranging from "C" to "E," and have a highrating variable equal to zero.

We include the highrating dummy, the CFPB consumer complaint measure, and their interaction term in the regression presented in Table X. Following the same specification as in Table III, we find that the positive relationship between distrust in traditional banks and P2P borrowing is driven by the low rated borrowers.

We continue our analysis with the specification from Table VI, but now try to identify which credit quality group is more sensitive to changes in interpersonal trust. We include the highrating dummy, the interpersonal trust measure, and their interaction term in the regression presented in Table XI. The coefficients from last column indicate that the negative relationship between higher interpersonal trust and P2P borrowing comes mostly from the low rated borrowers.

5.4.2 Joint impact of bank distrust, interpersonal trust, and Madoff scandal

To test whether bank misconduct and interpersonal trust are jointly influencing P2P borrowing, we include both regressors in the same specification, which yields the results presented in Table XII. The specifications are identical to those in Table III. We observe that both regressors remain statistically significant at the 1% level and the signs remain unchanged. Hence, each measure constitutes a substantial influence over the borrower's decision to obtain credit from an online platform. We also test whether interpersonal trust and Madoff victims in the state influence the online lending origination jointly, and then add the CFPB complaint measure. We find that the results from bank distrust, interpersonal trust and Madoff scandal are stable in Table XIII and Table XIV.

Overall, our empirical analysis indicates that trust in financial intermediation and interpersonal trust have shaped the development of online lending markets in the United States. The relationship is primarily driven by borrowers, rather than investors, and is most strongly influenced by borrowers with poor credit characteristics.

6 Conclusion

Trust has historically played a central role in the development of financial markets. In the absence of trust, borrower and investor apprehension prevents individuals from entering into mutually-beneficial contracts. Consequently, high trust societies benefit from greater financial depth and lower transaction costs. And low trust societies lack financial services and rely heavily on informal borrowing. This relationship between trust and financial markets may be especially relevant in the wake of the great financial crisis, which has given rise to the notion that "fraud has become a feature and not a bug" (Zingales (2015), p.19).

The existing literature focuses primarily on trust in traditional lending markets. Following Rau (2017), we examine evidence on the role of trust in online lending, which has grown rapidly in the U.S. since 2005; and consider multiple measures: interpersonal trust, trust in traditional banking, and trust in traditional finance. The first two operate primarily through borrowers. The last operates through investors.

To measure the expansion of U.S. online lending, we use loan book data and SEC filings. We also use 1) a survey-based measure of interpersonal trust from the GSS; 2) a measure of bank misconduct with both geographic and time variation, computed using the CFPB's database of consumer financial complaints; and 3) a general measure of distrust in traditional finance, captured by the geographic distribution of Madoff scandal victims. Our main regressions use data from LendingClub, which is the largest platform and has better data availability; however, our results are robust to the inclusion of Prosper data.

We first identify the impact of trust in traditional banking on the ratio of P2P debt to total debt. Our measure of trust is derived from consumer financial complaints submitted to the CFPB. We show that an increase of one complaint per bank branch is associated with a 6% increase in the ratio of online-to-total debt for the median state.

We also show that an increase in interpersonal trust is associated with a reduction in P2P debt relative to other debt. In particular, a one standard deviation increase in surveyreported interpersonal trust is associated with a -11% decrease in the online-to-total debt ratio for the median state. This suggests that in-person, bank-based borrowing and informal borrowing benefit from improvements in interpersonal trust; whereas online borrowing is largely perceived to be impersonal. We also show that interpersonal trust operates in the same direction on credit card debt, but in the opposite direction on mortgage debt. This aligns well with the observation that credit card debt is largely impersonal; whereas mortgage debt often requires personal interactions with bank employees.

In addition to our results for borrowers, we also consider a shock that affects only investors-namely, the number of Madoff scandal victims in a state-to discern the impact of an decline of trust in traditional finance. We find that investment in online lending is positively affected, which is consistent with the conjecture that the scandal shattered the confidence of wealthy investors, spurring interest in alternative investments outside the traditional finance sphere, such as online lending.

Overall, we find evidence that trust played a statistically and economically significant role in the expansion of online lending. Additionally, trust and factors that affect trust tended to push online lending and bank-based lending in opposite directions. Events that erode trust in bank-based borrowing, such as financial misconduct, increased online borrowing. Similarly, increases in interpersonal trust appear to benefit in-person, bank-based borrowing at the expense of online borrowing. Going forward, we expect that the dimensions of trust explored in this paper will remain important determinants of online lending growth and, more generally, FinTech development.

References

- Algan, Yann and Pierre Cahuc, "Inherited Trust and Growth," The American Economic Review, 2010, 100 (5), 2060–92.
- and __, "Trust, Growth, and Well-Being: New Evidence and Policy Implications," in "Handbook of Economic Growth," Vol. 2 2014, chapter 2, pp. 49–120.
- Balyuk, Tetyana, "Financial Innovation and Borrowers: Evidence from Peer-to-Peer Lending," *Mimeo*, 2016.
- Belleflamme, Paul, Nessrine Omrani, and Martin Peitz, "The Economics of Crowdfunding Platforms," *Mimeo*, 2015.
- Bertsch, Christoph, Isaiah Hull, and Xin Zhang, "Monetary Normalizations and Consumer Credit: Evidence from Fed Liftoff and Online Lending," *Mimeo*, 2017.
- Bester, Helmut, "Screening vs. rationing in credit markets with imperfect information," The American Economic Review, 1985, 75 (4), 850–855.
- Butler, Alexander W., Jess Cornaggia, and Umit G. Gurun, "Substitution between Sources of Finance in Consumer Capital Markets," *forthcoming in Management Science*, 2015.
- Chen, Ning, Arpita Ghosh, and Nicolas S. Lambert, "Auctions for social lending: A theoretical analysis," *Games and Economic Behavior*, 2014, *86*, 367–391.
- Corsetti, Giancarlo, Michael P. Devereux, Luigi Guiso, John Hassler, Gilles Saint-Paul, Hans-Werner Sinn, Jan-Egbert Sturm, and Xavier Vives, "A trust-driven financial crisis," *EEAG Report* on the European Economy, 2010.
- Crowe, Christopher and Rodney Ramcharan, "House Prices and Household Credit Access: Evidence from an Internet Bank," Journal of Money, Credit & Banking, 2013, 45 (6), 1085–1115.
- De Roure, Calebe, Loriana Pelizzon, and Paolo Tasca, "How does P2P lending fit into the consumer credit market? Calebe," *Deutsche Bundesbank Discussion Paper No. 30/2016*, 2016.
- Demyanyk, Yuliya, "Peer-to-Peer Lending Is Poised to Grow," Federal Reserve Bank of Cleveland, 2014.
- _, Elena Loutskina, and Daniel Kolliner, "The Taste of Peer-to-Peer Loans," Mimeo, 2017.
- Dohmen, Thomas, Armin Falk, David Huffman, and Uwe Sunde, "The Intergenerational Transmission of Risk and Trust Attitudes," *Review of Economic Studies*, 2012, 79 (2), 645–677.

- **Duarte, Jefferson, Stephan Siegel, and Lance Young**, "Trust and Credit: The Role of Appearance in Peer-to-peer Lending," *Review of Financial Studies*, 2012, 25 (8), 2455–2484.
- Einav, Liran, Mark Jenkins, and Jonathan Levin, "The Impact of Information Technology on Consumer Lending," *RAND Journal of Economics*, 2013, 44 (2), 249–274.
- FDIC, "Federal Deposit Insurance Corporation National Survey of Unbanked and Underbanked Households," 2015.
- Fehr, Ernst, "On the Economics and Biology of Trust," Journal of the European Economic Association, 2009, 7 (2-3), 235–266.
- FSB, FinTech credit Market structure, business models and financial stability implications 2017.
- Gambetta, Diego, "Can We Trust Trust?," in Diego Gambetta, ed., Trust: Making and Breaking Cooperative Relations, 2000, pp. 213–237.
- Giannetti, Mariassunta and Tracy Y. Wang, "Corporate Scandals and Household Stock Market Participation," Journal of Finance, 2016, 71 (6), 2591–2636.
- Gonzales, Laura and Yuliya Komarova Loureiro, "When Can a Photo Increase Credit?: The Impact of Lender and Borrower Profiles on Online P2P Loans," *Journal of Experimental and Behavorial Finance*, 2014, 2, 44–58.
- **Guiso, Luigi**, "A trust driven financial crisis: Implications for the future of financial markets," *EEAG* Report on the European Economy 2010, 2010.
- ____, Paola Sapienza, and Luigi Zingales, "The Role of Social Capital in Financial Development," American Economic Review, 2004, 94 (3), 526–556.
- _, _, and _, "Trusting the Stock Market," Journal of Finance, 2008, 63 (6), 2557–2600.
- _ , _ , and _ , "The Determinants of Attitudes towards Strategic Default on Mortgages," Journal of Finance, 2013, 68 (4), 1473–1515.
- Havrylchyk, Olena, Carlotta Mariotto, Talal Rahim, and Marianne Verdier, "What drives the expansion of the peer-to-peer lending?," *Mimeo*, 2017.
- Hertzberg, Andrew, Andres Liberman, and Daniel Paravisini, "Adverse Selection on Maturity: Evidence From on-Line Consumer Credit," *Mimeo*, 2015.
- Hildebrand, Thomas, Manju Puri, and Jörg Rocholl, "Adverse incentives in crowdfunding," Management Science, 2016, 63 (3), 587 608.

- Iyer, Rajkamal, Asim Ijaz Khwaja, Erzo F.P. Luttmer, and Kelly Shue, "Screening peers softly: inferring the quality of small borrowers," *Management Science*, 2015, 62 (6), 1554 – 1577.
- Lin, Mingfeng and Siva Viswanathan, "Home Bias in Online Investments: An Empirical Study of an Online Crowdfunding Market," *Management Science*, 2016, 62 (5), 1393 – 1414.
- Morse, Adair, "Peer-to-peer crowdfunding: Information and the potential for disruption in consumer lending," *NBER Working Paper No. 20899*, 2015, pp. 1–30.
- Paravisini, Daniel, Veronica Rappoport, and Enrichetta Ravina, "Risk aversion and wealth: evidence from person-to-person lending portfolios," *Management Science*, 2016, 63 (2), 279 297.
- Pope, Devin G. and Justin R. Sydnor, "What's in a Picture? Evidence of Discrimination from Prosper.com," The Journal of Human Resources, 2011, 46 (1), 53–92.
- **PricewaterhouseCoopers**, "Peer pressure: How peer-to-peer lending platforms are transforming the consumer lending industry," 2015, (February).
- Rajan, Raghuram G. and Luigi Zingales, "Financial Dependence and Growth," The American Economic Review, 1998, 88 (3), 559–586.
- Rau, Raghavendra, "Law, trust, and the development of crowdfunding," Mimeo, 2017.
- Ravina, Enrichetta, "Love & loans: the effect of beauty and personal characteristics in credit markets," *Mimeo*, 2012, pp. 1–79.
- Sakalauskait, Ieva, "Bank Risk-Taking and Misconduct," Mimeo, 2016.
- Senney, Garrett T., "The Geography of Bidder Behavior in Peer-to-Peer Lending Markets," Mimeo, 2016.
- Stevenson, Betsey and Justin Wolfers, "Trust in Public Institutions over the Business Cycle," The American Economic Review, 2011, 101 (3), 281–287.
- Tabellini, Guido, "Presidential address: Institutions and Culture," Journal of the European Economic Association, 2008, 6 (2-3), 255–294.
- Wei, Zaiyan and Mingfeng Lin, "Market Mechanisms in Online Crowdfunding," Mimeo, 2015.
- Xia, Lan, Kent B. Monroe, and Jennifer L. Cox, "The Price Is Unfair! A Conceptual Framework of Price Fairness Perceptions," *Journal of Marketing*, 2004, 68 (4), 1–15.
- Zhang, J. and P. Liu, "Rational Herding in Microloan Markets," *Management Science*, 2012, 58 (5), 892–912.
- Zingales, Luigi, "Presidential address: Does Finance Benefit Society?," The Journal of Finance, 2015, 70 (4), 1327–1363.

Variable	Description	Source			
State level					
CFPB complaints	Total number of consumer complaints regarding	Consumer Financial Pro-			
	banking services	tection Bureau (CFPB)			
Interpersonal trust	A survey-based measure of general trust in other peo-	General Social Survey			
	ple	(GSS)			
Madoff	Number of investors that suffered losses due to the	New York Times			
	Madoff Scandal				
Total debt	Total dollar amount of household debt; state level	NY Fed Consumer Panel			
Credit card debt	Total dollar amount of credit card debt; state level	NY Fed Consumer Panel			
P2P debt	Total dollar amount of P2P debt; state level	Platforms, computed by			
		authors			
GDP	Gross Domestic production in the past 12 months	US Census Bureau			
Population	Total number of population registered in the state in	Bureau of Labor Statistics			
	the past 12 months				
Total Area	Total land area of the state	US Census Bureau			
Branch No.	Total number of bank branches in the state	FDIC SoD data			
Loan-Borrower level					
Maturity	Maturity that is recorded under the loan identifica-	Platforms			
v	tion number				
Loan size	Dollar Amount of loan that is applied for	Platforms			
Loan interest rate	Interest rate that is assigned to the loan	Platforms			
Credit rating	Credit rating	Platforms			
Loan purpose	The purpose for the online lending loans	Borrowers			
Income	Annual income	Borrowers reported, veri-			
		fied by platforms			
Employment	Length of employment history	Borrowers reported, veri-			
		fied by platforms			

Table I: Variable definitions

	100010 110	s annai j	5000150105				
	Mean	SD	P25	Median	P75	Ν	
State-level variables (by state)							
Interpersonal trust	0.40	0.11	0.33	0.39	0.46	49	
Madoff in thousands	0.86	1.15	0.17	0.40	1.30	51	
State-level variables (by state and year)							
Credit card debt $(10b\$)$	1.53	1.83	0.38	0.91	1.94	459	
Total debt $(10b\$)$	24.35	33.20	6.04	15.19	32.06	459	
population density	0.11	0.43	0.02	0.03	0.07	459	
Log GDP	12.21	1.01	11.43	12.29	12.94	459	
Log population	14.94	1.03	14.20	15.08	15.64	459	
Unemployment rate	6.76	2.12	5.10	6.60	8.10	459	
Number of bank branches (k)	1.82	1.73	0.47	1.46	2.38	459	
State-level variables (by state an	nd month)						
CFPB complaints per branch	0.20	0.66	0.06	0.10	0.16	2839	
P2P debt (m\$)	5.83	11.41	0.31	1.59	6.32	4653	
P2P debt/Total debt	0.24	0.25	0.02	0.15	0.41	4653	
Average DTI ratio	0.17	0.06	0.13	0.17	0.19	4653	
Average interest rate	0.13	0.01	0.13	0.13	0.14	4653	
Average annual income (k\$)	75.08	26.41	65.78	73.16	81.56	4653	
Average employment duration	5.30	1.11	4.95	5.50	5.82	4653	
Loan-borrower level variables							
Credit card debt (per cap)	2947.62	566.29	2515.00	2880.00	3350.00	459	
Total debt (per cap)	46358.24	12645.88	36595.00	43295.00	54255.00	459	
Loan size	14961.15	8791.76	8000.00	13000.00	20000.00	1745948	
Interest rate	0.14	0.05	0.10	0.13	0.16	1745948	
High rating	0.37	0.48	0.00	0.00	1.00	1745948	
Annual income (k\$)	79.72	180.73	45.00	65.00	90.00	1745948	
Employment	5.56	3.79	2.00	5.00	10.00	1745948	

Table II: Summary statistics

Notes: This table shows the summary statistics for all variables used in the empirical analysis. The sample covers the largest P2P platform, LendingClub, between 2008 and 2016. Where possible, we use state-level variables with monthly frequencies, while the loan-borrower level variables contain individual-specific information. The CFPB data extends back to 2012. Thus, the observation number is smaller than for other variables. It also means that the loan level regression with the CFPB complaints measure will use a fraction of the whole application sample. The variable *interpersonal trust* is computed as the state-level average of positive responses to the General Social Survey's (GSS) question about interpersonal trust over the 1973-2006 period, however there are only 49 states in our sample. Note that the Madoff victim count is 51 because we do not include U.S. territories. The variable *Credit card debt* is the state-level total credit card debt, and *total debt* is the state total household debt not channeled through online lending. Both are in 10-Billion dollar terms, provided by New York Fed consumer panel dataset.

			0			1		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
CFPB	0.017^{**} (0.007)	$\begin{array}{c} 0.021^{***} \\ (0.005) \end{array}$	$\begin{array}{c} 0.023^{***} \\ (0.007) \end{array}$	$\begin{array}{c} 0.012^{***} \\ (0.002) \end{array}$	$\begin{array}{c} 0.009^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.008^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.009^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.009^{***} \\ (0.001) \end{array}$
DTI		0.669^{*} (0.366)		0.408^{*} (0.216)		0.284^{*} (0.150)		$\begin{array}{c} 0.113^{***} \\ (0.038) \end{array}$
interest rate		-4.054^{***} (1.413)		-2.493^{**} (1.091)		-0.814 (1.078)		1.300^{*} (0.712)
income		$\begin{array}{c} 0.003^{***} \\ (0.001) \end{array}$		0.001^{**} (0.001)		$0.000 \\ (0.000)$		-0.000 (0.000)
employment		-0.000 (0.013)		-0.001 (0.009)		$0.004 \\ (0.009)$		0.013^{*} (0.007)
pop. density		$\begin{array}{c} 0.007 \\ (0.012) \end{array}$		-1.299^{***} (0.185)		-1.195^{***} (0.164)		-1.141^{***} (0.163)
log GDP		-0.049^{**} (0.020)		-0.546^{***} (0.140)		-0.404^{***} (0.138)		-0.442^{***} (0.139)
log population		$\begin{array}{c} 0.076^{***} \\ (0.018) \end{array}$		$1.746^{***} \\ (0.564)$		-0.293^{*} (0.156)		-0.213 (0.145)
unemployment rate		-0.052^{***} (0.007)		-0.129^{***} (0.010)		-0.024^{***} (0.003)		-0.024^{***} (0.003)
Year FE	NO	NO	NO	NO	YES	YES	NO	NO
State FE	NO	NO	YES	YES	YES	YES	YES	YES
Year-Month FE	NO	NO	NO	NO	NO	NO	YES	YES
Controls	NO	YES	NO	YES	NO	YES	NO	YES
No. of observations	2839	2839	2839	2839	2839	2839	2839	2839
Adj. R-square	0.002	0.270	0.111	0.567	0.684	0.692	0.874	0.881

Table III: Online lending and CFPB consumer complaints

$$Y_{i,t} = \beta_1 \text{CFPB complaints}_{i,t} + \gamma X_{i,t} + (A_i + B_t) + \epsilon_{i,t}$$

using the CFPB consumer complaints as the key explanatory variable to measure consumers' distrust in traditional banking. The dependent variable is the fraction of online lending demand compared to the household total debt at the monthly frequency. *CFPB* is the number of consumer complaints per branch in state *i* in month *t*. We include a number of independent variables to control for the average quality of loans. *DTI* is the simple average debt-to-income ratio of all the loans from state *i* at month *t*. *Interest rate* is the average value of interest rates for loans in the state at month *t*. *Income* and *Employment* measure the average annual income and years of employment for the borrowers in state *i* in month *t*. We also control for state level variables such as population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population and state unemployment rate. The different columns present different combinations of state fixed effects and time fixed effects. Standard errors, in parentheses, are corrected for clustering of observations by year-month. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
CFPB	$\begin{array}{c} 0.071^{***} \\ (0.022) \end{array}$	0.079^{***} (0.016)	$\begin{array}{c} 0.063^{***} \\ (0.019) \end{array}$	$\begin{array}{c} 0.030^{***} \\ (0.005) \end{array}$	$\begin{array}{c} 0.020^{***} \\ (0.004) \end{array}$	$\begin{array}{c} 0.020^{***} \\ (0.004) \end{array}$	$\begin{array}{c} 0.020^{***} \\ (0.003) \end{array}$	$\begin{array}{c} 0.019^{***} \\ (0.003) \end{array}$
DTI		1.527^{*} (0.903)		1.055^{*} (0.561)		0.720^{*} (0.390)		$\begin{array}{c} 0.218^{**} \\ (0.105) \end{array}$
interest rate		-12.974^{***} (4.053)		-7.828^{**} (3.081)		-3.360 (3.070)		$1.267 \\ (1.863)$
income		0.008^{***} (0.002)		$\begin{array}{c} 0.002 \\ (0.001) \end{array}$		-0.000 (0.001)		-0.001 (0.001)
employment		$\begin{array}{c} 0.002 \\ (0.039) \end{array}$		$\begin{array}{c} 0.003 \\ (0.024) \end{array}$		$0.015 \\ (0.022)$		0.038^{**} (0.017)
pop. density		-0.058 (0.041)		-1.516^{***} (0.471)		-1.235^{***} (0.404)		-1.155^{***} (0.378)
log GDP		$\begin{array}{c} 0.517^{***} \\ (0.064) \end{array}$		-1.508^{***} (0.380)		-1.168^{***} (0.364)		-1.273^{***} (0.368)
log population		-0.420^{***} (0.062)		7.300^{***} (1.687)		$\frac{1.869^{***}}{(0.416)}$		$2.078^{***} \\ (0.437)$
unemployment rate		-0.150^{***} (0.018)		-0.371^{***} (0.028)		-0.100^{***} (0.009)		-0.100^{***} (0.009)
Year FE	NO	NO	NO	NO	YES	YES	NO	NO
State FE	NO	NO	YES	YES	YES	YES	YES	YES
Year-Month FE	NO	NO	NO	NO	NO	NO	YES	YES
Controls	NO	YES	NO	YES	NO	YES	NO	YES
No. of observations	2839	2839	2839	2839	2839	2839	2839	2839
Adj. R-square	0.005	0.324	0.119	0.627	0.722	0.730	0.891	0.896

Table IV: Extensive margin: number of applicants/population and CFPB complaints

Notes: This table reports the extensive margin regression:

 $Y_{i,t} = \beta_1 \text{CFPB complaints}_{i,t} + \gamma X_{i,t} + (A_i + B_t) + \epsilon_{i,t}$

using the CFPB consumer complaints as the key explanatory variable to measure consumers' distrust in traditional banking. The dependent variable is the fraction (basis points) of online lending applicants in the state population at month t. CFPB is the number of consumer complaints per branch in state i in month t. We include a number of independent variables to control for the average quality of loans. DTI is the simple average debt-to-income ratio of all the loans in state i in month t. Interest rate is the average value of interest rates for loans in state i in month t. Income and Employment measure the average annual income and years of employment for the borrowers in state i in month t. We also control for state level variables such as population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population and state unemployment rate. The different columns present different combinations of state fixed effects and time fixed effects. Standard errors, in parentheses, are corrected for clustering of observations by year-month. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
CFPB	$0.003 \\ (0.007)$	$0.005 \\ (0.007)$	$0.000 \\ (0.002)$	$0.000 \\ (0.001)$	-0.001^{*} (0.000)	-0.000 (0.000)	-0.000 (0.000)	-0.000 (0.000)
homeowner		$\begin{array}{c} 0.047^{***} \\ (0.005) \end{array}$		$\begin{array}{c} 0.019^{***} \\ (0.003) \end{array}$		$\begin{array}{c} 0.019^{***} \\ (0.003) \end{array}$		$\begin{array}{c} 0.019^{***} \\ (0.003) \end{array}$
has mortgage		$\begin{array}{c} 0.034^{***} \\ (0.003) \end{array}$		$\begin{array}{c} 0.037^{***} \\ (0.002) \end{array}$		$\begin{array}{c} 0.037^{***} \\ (0.002) \end{array}$		$\begin{array}{c} 0.037^{***} \\ (0.002) \end{array}$
employment		$\begin{array}{c} 0.003^{***} \\ (0.000) \end{array}$		$\begin{array}{c} 0.004^{***} \\ (0.000) \end{array}$		$\begin{array}{c} 0.004^{***} \\ (0.000) \end{array}$		$\begin{array}{c} 0.004^{***} \\ (0.000) \end{array}$
annual income		0.000^{***} (0.000)		0.000^{***} (0.000)		0.000^{***} (0.000)		0.000^{***} (0.000)
interest rate		$\begin{array}{c} 0.316^{***} \\ (0.025) \end{array}$		$\begin{array}{c} 0.280^{***} \\ (0.020) \end{array}$		0.290^{***} (0.017)		$\begin{array}{c} 0.304^{***} \\ (0.017) \end{array}$
maturity		$\begin{array}{c} 0.074^{***} \\ (0.004) \end{array}$		$\begin{array}{c} 0.074^{***} \\ (0.004) \end{array}$		$\begin{array}{c} 0.074^{***} \\ (0.004) \end{array}$		$\begin{array}{c} 0.073^{***} \\ (0.004) \end{array}$
Year FE	NO	NO	NO	NO	YES	YES	NO	NO
State FE	NO	NO	YES	YES	YES	YES	YES	YES
Year-Month FE	NO	NO	NO	NO	NO	NO	YES	YES
Controls	NO	YES	NO	YES	NO	YES	NO	YES
No. of observations	1680778	1680778	1680778	1680778	1680778	1680778	1680778	1680778
Adj. R-square	0.000	0.236	0.138	0.306	0.140	0.306	0.142	0.309

Table V: Intensive margin: size of loan request and CFPB complaints

Notes: This table reports the extensive margin regression:

 $Y_{i,j,t} = \beta_1 \text{CFPB complaints}_{i,t} + \gamma X_{i,j,t} + (A_i + B_t) + \epsilon_{i,j,t}$

using the CFPB consumer complaints as the key explanatory variable to measure consumers' distrust in traditional banking. The dependent variable is the fraction of applicant j's online loan request amount compared to the per capita household total debt in state i in month t. *CFPB* is the number of consumer complaints per branch in state i in month t. We include a number of independent variables to control for the individual borrower's characteristics. homeowner and has mortgage are dummy variables about the applicant's homeownership status. employment and annual income indicate the applicant's job and income situation. We also control for applicants' FICO score and loan request characteristics like the interest rate and the maturity. The different columns here present different combinations of state fixed effects and time fixed effects. Standard errors, in parentheses, are corrected for clustering of observations by state. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)
interpersonal trust	-0.103^{***} (0.024)	-0.346^{***} (0.057)	-0.133^{***} (0.024)	-0.154^{***} (0.025)	-0.134^{***} (0.024)	-0.150^{***} (0.024)
DTI		$1.125^{**} \\ (0.436)$		$0.194 \\ (0.127)$		$\begin{array}{c} 0.074 \\ (0.047) \end{array}$
interest rate		$0.080 \\ (0.697)$		-0.121 (0.523)		$\begin{array}{c} 0.570 \ (0.437) \end{array}$
income		0.000^{*} (0.000)		$0.000 \\ (0.000)$		-0.000 (0.000)
employment		$\begin{array}{c} 0.021^{***} \\ (0.004) \end{array}$		$0.001 \\ (0.003)$		$0.003 \\ (0.003)$
pop. density		0.020^{**} (0.010)		-0.045^{***} (0.007)		-0.042^{***} (0.006)
log GDP		$\begin{array}{c} 0.054^{***} \\ (0.019) \end{array}$		$\begin{array}{c} 0.027^{***} \\ (0.007) \end{array}$		$\begin{array}{c} 0.022^{***} \\ (0.007) \end{array}$
log population		$\begin{array}{c} 0.012 \\ (0.020) \end{array}$		-0.030^{***} (0.007)		-0.026^{***} (0.006)
unemployment rate		-0.063^{***} (0.007)		$\begin{array}{c} 0.004^{***} \\ (0.001) \end{array}$		$\begin{array}{c} 0.003^{***} \\ (0.001) \end{array}$
Branch		-0.014^{***} (0.002)		0.003^{**} (0.001)		0.003^{**} (0.001)
Year FE	NO	NO	YES	YES	NO	NO
State FE	NO	NO	NO	NO	NO	NO
Year-Month FE	NO	NO	NO	NO	YES	YES
Controls	NO	YES	NO	YES	NO	YES
No. of observations	4297	4297	4297	4297	4297	4297
Adj. R-square	0.002	0.466	0.760	0.765	0.866	0.870

Table VI: Online lending and interpersonal trust

Notes: This table reports the regression on:

 $Y_{i,t} = \beta_1 \text{Interpersonal trust}_i + \gamma X_{i,t} + (B_t) + \epsilon_{i,t}.$

 $Y_{i,t}$ is the total amount of online lending demand in state *i* in month *t* normalized by total bank debt. Interpersonal trust is the average of the social trust question responses in the General Social Survey (GSS) over the period 1973-2006. We include a number of independent variable to control for the average quality of loans. *DTI* is the simple average debt-to-income ratio of all the loans from in state *i* in month *t*. Interest rate is the average value of interest rates for loans in state *i* in month *t*. Interest rate is the average value of interest rates for loans in state *i* in month *t*. We also control for state level variables such as population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population, state unemployment rate, and the number of bank branches in thousands. Standard errors, in parentheses, are corrected for clustering of observations by year-month. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level, respectively.

	(1)	(2)	(3)	(4)
interpersonal trust	-0.004^{***} (0.001)	-0.010^{**} (0.004)	-0.004^{***} (0.001)	-0.014^{***} (0.002)
pop. density		-0.002 (0.002)		-0.003^{***} (0.001)
log GDP		-0.018^{**} (0.005)		-0.013^{***} (0.003)
log population		0.013^{**} (0.004)		0.009^{***} (0.002)
unemployment rate		-0.000 (0.001)		-0.001^{***} (0.000)
Branch		$\begin{array}{c} 0.003^{***} \\ (0.001) \end{array}$		$\begin{array}{c} 0.002^{***} \\ (0.001) \end{array}$
Year FE	NO	NO	YES	YES
State FE	NO	NO	NO	NO
Year-Month FE	NO	NO	NO	NO
Controls	NO	YES	NO	YES
No. of observations	441	441	441	441
Adj. R-square	0.001	0.254	0.238	0.431

Table VII: Credit card debt and interpersonal trust

Notes: This table reports the robustness regression as:

 $Y_{i,t} = \beta_1 \text{Interpersonal trust}_i + \gamma X_{i,t} + (B_t) + \epsilon_{i,t}.$

 $Y_{i,t}$ is the fraction of aggregate outstanding credit card debt in the total outstanding bank debt to households in state *i* in year *t*. Interpersonal trust is the average of the social trust answers in the General Social Survey (GSS) over the period 1973-2006. We also control for state level variables such as population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population, the state unemployment rate, and the number of bank branches in thousands. Standard errors, in parentheses, are corrected for clustering of observations by year. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level, respectively.

	(1)	(2)	(3)	(4)
interpersonal trust	$\begin{array}{c} 0.126^{***} \\ (0.009) \end{array}$	$\begin{array}{c} 0.185^{***} \\ (0.014) \end{array}$	$\begin{array}{c} 0.126^{***} \\ (0.009) \end{array}$	$\begin{array}{c} 0.168^{***} \\ (0.004) \end{array}$
pop. density		-0.022^{*} (0.011)		-0.029^{***} (0.002)
log GDP		$\begin{array}{c} 0.127^{***} \\ (0.020) \end{array}$		$\begin{array}{c} 0.151^{***} \\ (0.006) \end{array}$
log population		-0.115^{***} (0.023)		-0.132^{***} (0.008)
unemployment rate		$\begin{array}{c} 0.013^{***} \\ (0.004) \end{array}$		$\begin{array}{c} 0.010^{***} \\ (0.001) \end{array}$
Branch		-0.007^{**} (0.002)		-0.010^{***} (0.002)
Year FE	NO	NO	YES	YES
State FE	NO	NO	NO	NO
Year-Month FE	NO	NO	NO	NO
Controls	NO	YES	NO	YES
No. of observations	441	441	441	441
Adj. R-square	0.047	0.334	0.179	0.454

Table VIII: Mortgage lending and interpersonal trust

Notes: This table reports the robustness regression as:

 $Y_{i,t} = \beta_1 \text{Interpersonal trust}_i + \gamma X_{i,t} + (B_t) + \epsilon_{i,t}.$

 $Y_{i,t}$ is the fraction of aggregate outstanding mortgage debt in the total outstanding bank debt to households in state *i* in year *t*. Interpersonal trust is the average of the social trust question responses in the General Social Survey (GSS) over the period 1973-2006. We also control for state level variables such as population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population, the state unemployment rate, and the number of bank branches in thousands. Standard errors, in parentheses, are corrected for clustering of observations by year. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)
Madoff victims	0.001 (0.000)	-0.005^{***} (0.001)	$\begin{array}{c} 0.004^{***} \\ (0.001) \end{array}$	0.006^{***} (0.001)	$\begin{array}{c} 0.004^{***} \\ (0.001) \end{array}$	0.006^{***} (0.001)
DTI		$\begin{array}{c} 0.969^{***} \\ (0.262) \end{array}$		$0.068 \\ (0.056)$		$\begin{array}{c} 0.016 \\ (0.030) \end{array}$
interest rate		-0.087 (0.496)		-0.119 (0.261)		$\begin{array}{c} 0.232^{***} \\ (0.086) \end{array}$
income		$\begin{array}{c} 0.001^{***} \\ (0.000) \end{array}$		$0.000 \\ (0.000)$		0.000^{*} (0.000)
employment		$\begin{array}{c} 0.143^{***} \\ (0.021) \end{array}$		0.010^{**} (0.004)		$0.004 \\ (0.003)$
pop. density		$\begin{array}{c} 0.024^{***} \\ (0.008) \end{array}$		-0.023^{***} (0.003)		-0.022^{***} (0.003)
log GDP		0.037^{**} (0.018)		-0.003 (0.005)		-0.006 (0.005)
log population		$\begin{array}{c} 0.013 \\ (0.019) \end{array}$		-0.005 (0.005)		-0.003 (0.005)
unemployment rate		-0.042^{***} (0.005)		0.006^{***} (0.001)		0.006^{***} (0.001)
Branch		-0.016^{***} (0.002)		0.002^{**} (0.001)		0.002^{**} (0.001)
Year FE	NO	NO	YES	YES	NO	NO
State FE	NO	NO	NO	NO	NO	NO
Year-Month FE	NO	NO	NO	NO	YES	YES
Controls	NO	YES	NO	YES	NO	YES
No. of observations	4653	4653	4653	4653	4653	4653
Adj. R-square	0.000	0.434	0.826	0.829	0.930	0.933

Table IX: Online lending and Madoff scandal victims

 $Y_{i,t} = \beta_1 \text{Madoff victims}_i + \gamma X_{i,t} + (B_t) + \epsilon_{i,t}.$

The dependent variable is the total amount of originated online lending in state i in month t normalized by total bank debt. *Madoff victims* is the number (in thousands) of investors who suffered losses in the scandal at the state level. We include a number of independent variables to control for the average quality of loans. *DTI* is the simple average debt-to-income ratio of all the loans from in state i in month t. *Interest rate* is the average value of interest rates of loans in state i in month t. *Interest rate* is the average value of and years of employment for the borrowers in state i in month t. We also control for state level variables such as population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population, state unemployment rate, and the number of bank branches in thousands. Standard errors, in parentheses, are corrected for clustering of observations by year-month. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
CFPB	0.010^{**} (0.004)	$\begin{array}{c} 0.012^{***} \\ (0.003) \end{array}$	$\begin{array}{c} 0.014^{***} \\ (0.004) \end{array}$	0.008^{***} (0.001)	$\begin{array}{c} 0.007^{***} \\ (0.001) \end{array}$				
high rating=1	-0.073^{***} (0.006)	-0.078^{***} (0.006)	-0.073^{***} (0.006)	-0.078^{***} (0.006)	-0.073^{***} (0.006)	-0.075^{***} (0.006)	-0.073^{***} (0.006)	-0.075^{***} (0.006)	
high rating=1 x CFPB	-0.005^{***} (0.002)	-0.004^{**} (0.002)	-0.005^{***} (0.002)	-0.004^{**} (0.002)	-0.005^{***} (0.002)	-0.005^{***} (0.002)	-0.005^{***} (0.002)	-0.005^{***} (0.002)	
interest rate		-0.144^{***} (0.041)		-0.122^{***} (0.033)		-0.061^{**} (0.026)		-0.035^{*} (0.018)	
income		$0.000 \\ (0.000)$		-0.000 (0.000)		-0.000 (0.000)		$0.000 \\ (0.000)$	
employment		-0.000 (0.000)		$0.000 \\ (0.000)$		$0.000 \\ (0.000)$		-0.000 (0.000)	
pop. density		$\begin{array}{c} 0.014^{***} \\ (0.004) \end{array}$		-0.605^{***} (0.090)		-0.587^{***} (0.081)		-0.587^{***} (0.081)	
$\log \text{GDP}$		-0.019^{***} (0.006)		-0.272^{***} (0.070)		-0.195^{***} (0.066)		-0.199^{***} (0.066)	
log population		0.036^{***} (0.007)		1.026^{***} (0.308)		-0.150^{**} (0.071)		-0.144^{**} (0.070)	
unemployment rate		-0.035^{***} (0.004)		-0.070^{***} (0.005)		-0.012^{***} (0.001)		-0.012^{***} (0.001)	
Year FE	NO	NO	NO	NO	YES	YES	NO	NO	
State FE	NO	NO	YES	YES	YES	YES	YES	YES	
Year-Month FE	NO	NO	NO	NO	NO	NO	YES	YES	
Controls	NO	YES	NO	YES	NO	YES	NO	YES	
No. of observations	5644	5644	5644	5644	5644	5644	5644	5644	
Adj. R-square	0.085	0.272	0.176	0.558	0.678	0.682	0.843	0.847	

Table X: Online lending and CFPB complaints: by loan quality

Notes: This table reports the regression of the online lending amount by borrower credit quality in a certain state for a given month, normalized by the total bank debt in the same state, on the credit rating dummy, CFPB consumer complaints, and their interactions. highrating=1 means the borrowers have a credit rating of either "A" or "B". *CFPB* is the number of consumer complaints per branch in state *i* in month *t*. We include a number of independent variables to control for the average quality of loans. *DTI* is the simple average debt-to-income ratio of all the loans from state *i* in month *t*. Interest rate is the average value of interest rates for loans in state *i* in month *t*. Income and Employment measure the average annual income and years of employment for the borrowers in state *i* in month *t*. We also control for state level variables such as population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population and state unemployment rate. The different columns present different combinations of state fixed effects and time fixed effects. Standard errors, in parentheses, are corrected for clustering of observations by year-month. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level, respectively.

	0	1		v	1 0	
	(1)	(2)	(3)	(4)	(5)	(6)
interpersonal trust	-0.069***	-0.232***	-0.081***	-0.094***	-0.081***	-0.094***
	(0.015)	(0.028)	(0.015)	(0.015)	(0.015)	(0.015)
highrating=1	-0.059***	-0.061***	-0.063***	-0.065***	-0.063***	-0.065***
	(0.008)	(0.009)	(0.008)	(0.008)	(0.008)	(0.008)
high rating=1 x interpersonal trust	0.034^{***}	0.037^{***}	0.042^{***}	0.043^{***}	0.042^{***}	0.043^{***}
	(0.010)	(0.011)	(0.010)	(0.010)	(0.010)	(0.010)
interest rate		0.007		-0.047^{**}		-0.030
		(0.042)		(0.022)		(0.019)
income		-0.000		-0.000		-0.000
		(0.000)		(0.000)		(0.000)
employment		0.002***		0.000^{*}		0.000**
		(0.000)		(0.000)		(0.000)
pop. density		0.007		-0.022***		-0.022***
		(0.006)		(0.003)		(0.003)
log GDP		0.020**		0.009***		0.009***
		(0.010)		(0.003)		(0.003)
log population		0.012		-0.012***		-0.012***
		(0.010)		(0.003)		(0.003)
unemployment rate		-0.041***		0.002***		0.001***
		(0.003)		(0.000)		(0.000)
branch		-0.000***		0.000***		0.000***
		(0.000)		(0.000)		(0.000)
Year FE	NO	NO	YES	YES	NO	NO
State FE	NO	NO	NO	NO	NO	NO
Year-Month FE	NO	NO	NO	NO	YES	YES
Controls	NO	YES	NO	YES	NO	YES
No. of observations	8406	8406	8406	8406	8406	8406
Adj. R-square	0.033	0.408	0.736	0.741	0.836	0.840

Table XI: Online lending and interpersonal trust: by loan quality

Notes: This table reports the regression of online lending amount by borrower credit quality in a certain state for a given month, normalized by the total debt in the same state, on the credit rating dummy, interpersonal trust, and their interactions. highrating=1 means the borrowers have a credit rating of either "A' 'or "B". Interpersonal trust is the average of the social trust question responses in the General Social Survey (GSS) over the period 1973-2006. We include a number of independent variables to control for the average quality of loans. DTI is the simple average debt-to-income ratio of all the loans from state *i* in month *t*. Interest rate is the average value of interest rates for loans in state *i* in month *t*. Income and Employment measure the average annual income and years of employment for the borrowers in state *i* in month *t*. We also control for state level variables such as population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population, state unemployment rate, and the number of bank branches in thousands. Standard errors, in parentheses, are corrected for clustering of observations by year-month. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level.

	(1)	(2)	(3)	(4)	(5)	(6)
interpersonal trust	-0.179^{***} (0.032)	-0.583^{***} (0.063)	-0.196^{***} (0.033)	-0.230^{***} (0.036)	-0.197^{***} (0.033)	-0.208^{***} (0.036)
CFPB	0.015^{**} (0.006)	$\begin{array}{c} 0.021^{***} \\ (0.004) \end{array}$	$\begin{array}{c} 0.007^{***} \\ (0.002) \end{array}$	$\begin{array}{c} 0.007^{***} \\ (0.002) \end{array}$	$\begin{array}{c} 0.007^{***} \\ (0.002) \end{array}$	$\begin{array}{c} 0.008^{***} \\ (0.002) \end{array}$
DTI		0.539^{*} (0.294)		$\begin{array}{c} 0.224 \\ (0.153) \end{array}$		$\begin{array}{c} 0.081 \\ (0.051) \end{array}$
interest rate		-4.168^{***} (1.370)		-0.606 (1.288)		1.247 (1.213)
income		$\begin{array}{c} 0.002^{***} \\ (0.001) \end{array}$		$0.000 \\ (0.001)$		-0.000 (0.000)
employment		$0.007 \\ (0.013)$		$0.007 \\ (0.012)$		$0.015 \\ (0.011)$
pop. density		-0.020 (0.014)		-0.068^{***} (0.010)		-0.060^{***} (0.010)
log GDP		$\begin{array}{c} 0.013 \ (0.017) \end{array}$		$\begin{array}{c} 0.040^{***} \\ (0.012) \end{array}$		$\begin{array}{c} 0.036^{***} \\ (0.012) \end{array}$
log population		$0.017 \\ (0.016)$		-0.039^{***} (0.012)		-0.034^{***} (0.012)
unemployment rate		-0.069^{***} (0.007)		0.006^{***} (0.002)		0.006^{***} (0.002)
Year FE	NO	NO	YES	YES	NO	NO
State FE	NO	NO	NO	NO	NO	NO
Year-Month FE	NO	NO	NO	NO	YES	YES
Controls	NO	YES	NO	YES	NO	YES
No. of observations	2751	2751	2751	2751	2751	2751
Adj. R-square	0.008	0.346	0.580	0.593	0.766	0.779

Table XII: Online lending, interpersonal trust and CFPB complaints

 $Y_{i,t} = \beta_1 \text{CFPB complaints}_{i,t} + \beta_2 \text{Interpersonal trust}_i + \gamma X_{i,t} + (B_t) + \epsilon_{i,t}$

using the CFPB consumer complaints as the key explanatory variable to measure consumers' distrust in traditional banking. The dependent variable is the fraction of online lending demand compared to the household total debt at the monthly frequency. *CFPB* is the number of consumer complaints per branch in state *i* in month *t*. *Interpersonal trust* is the measure of social trust from the General Social Survey (GSS) during 1973-2006. Variables *DTI*, *Interest rate*, *Income*, and *Employment* are the average measures for borrowers in state *i* in month *t*. We also control for state level variables such as population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population and state unemployment rate. Standard errors, in parentheses, are corrected for clustering of observations by year-month. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)
interpersonal trust	-0.027^{**} (0.012)	-0.195^{***} (0.030)	-0.031^{***} (0.009)	-0.035^{***} (0.009)	-0.032^{***} (0.008)	-0.034^{***} (0.007)
Madoff victims	$0.000 \\ (0.001)$	-0.008^{***} (0.001)	$\begin{array}{c} 0.004^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.005^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.004^{***} \\ (0.001) \end{array}$	$\begin{array}{c} 0.005^{***} \\ (0.001) \end{array}$
DTI		0.900^{***} (0.246)		$\begin{array}{c} 0.054 \\ (0.051) \end{array}$		$0.009 \\ (0.027)$
interest rate		-0.223 (0.495)		-0.136 (0.262)		$\begin{array}{c} 0.214^{***} \\ (0.071) \end{array}$
income		0.001^{***} (0.000)		$0.000 \\ (0.000)$		0.000^{*} (0.000)
employment		$\begin{array}{c} 0.146^{***} \\ (0.021) \end{array}$		$\begin{array}{c} 0.011^{***} \\ (0.004) \end{array}$		0.005^{*} (0.003)
pop. density		$\begin{array}{c} 0.021^{***} \\ (0.008) \end{array}$		-0.022^{***} (0.003)		-0.021^{***} (0.003)
log GDP		0.041^{**} (0.018)		-0.002 (0.004)		-0.004 (0.005)
log population		$0.007 \\ (0.018)$		-0.006 (0.005)		-0.004 (0.005)
unemployment rate		-0.048^{***} (0.005)		$\begin{array}{c} 0.004^{***} \\ (0.001) \end{array}$		0.003^{***} (0.001)
Branch		-0.011^{***} (0.002)		0.003^{***} (0.001)		0.003^{***} (0.001)
Year FE	NO	NO	YES	YES	NO	NO
State FE	NO	NO	NO	NO	NO	NO
Year-Month FE	NO	NO	NO	NO	YES	YES
Controls	NO	YES	NO	YES	NO	YES
No. of observations	4523	4523	4523	4523	4523	4523
Adj. R-square	0.001	0.455	0.829	0.832	0.936	0.938

Table XIII: Online lending, interpersonal trust and Madoff victims

 $Y_{i,t} = \beta_1 \text{Interpersonal trust}_i + \beta_2 \text{Madoff victims}_i + \gamma X_{i,t} + (B_t) + \epsilon_{i,t}.$

The dependent variable is the originated P2P debt amount in state i in month t normalized by total bank debt. *CFPB* is the number of consumer complaints per branch. *Interpersonal trust* is the measure of social trust from the General Social Survey (GSS) during 1973-2006. *Madoff victims* is the number (in thousands) of investors who suffered losses in the scandal. Variables *DTI*, *Interest rate*, *Income*, and *Employment* are the average measures for borrowers in state i in month t. We also control for state level variables such as population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population and state unemployment rate, and the number of bank branches in thousands. Standard errors, in parentheses, are corrected for clustering of observations by year-month. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level.

	(1)	(2)	(3)	(4)	(5)	(6)
CFPB	0.012^{**} (0.005)	$\begin{array}{c} 0.018^{***} \\ (0.003) \end{array}$	0.003^{**} (0.001)	0.003^{**} (0.001)	0.002^{**} (0.001)	0.003^{**} (0.001)
interpersonal trust	-0.039^{**} (0.016)	-0.349^{***} (0.034)	-0.045^{***} (0.013)	-0.042^{**} (0.018)	-0.047^{***} (0.013)	-0.027^{**} (0.012)
Madoff victims	0.004^{***} (0.001)	$0.000 \\ (0.002)$	$\begin{array}{c} 0.007^{***} \\ (0.001) \end{array}$	0.009^{***} (0.001)	$\begin{array}{c} 0.007^{***} \\ (0.001) \end{array}$	0.009^{***} (0.001)
DTI		$\begin{array}{c} 0.384^{**} \\ (0.153) \end{array}$		$0.066 \\ (0.064)$		$\begin{array}{c} 0.015 \\ (0.036) \end{array}$
interest rate		$\begin{array}{c} -4.617^{***} \\ (1.111) \end{array}$		-0.693 (0.840)		$\begin{array}{c} 0.687^{**} \\ (0.285) \end{array}$
income		$\begin{array}{c} 0.004^{***} \\ (0.001) \end{array}$		$0.000 \\ (0.000)$		0.001^{***} (0.000)
employment		$\begin{array}{c} 0.176^{***} \\ (0.047) \end{array}$		$\begin{array}{c} 0.089^{***} \\ (0.027) \end{array}$		0.041^{**} (0.016)
pop. density		$0.017 \\ (0.011)$		-0.030^{***} (0.005)		-0.029^{***} (0.004)
log GDP		-0.071^{***} (0.017)		-0.008 (0.013)		-0.018^{*} (0.010)
log population		$\begin{array}{c} 0.089^{***} \\ (0.016) \end{array}$		$0.001 \\ (0.012)$		$0.011 \\ (0.010)$
unemployment rate		-0.062^{***} (0.005)		0.006^{***} (0.001)		0.006^{***} (0.001)
Year FE	NO	NO	YES	YES	NO	NO
State FE	NO	NO	NO	NO	NO	NO
Year-Month FE	NO	NO	NO	NO	YES	YES
Controls	NO	YES	NO	YES	NO	YES
No. of observations	2739	2739	2739	2739	2739	2739
Adj. R-square	0.003	0.452	0.701	0.710	0.888	0.896

Table XIV: Online lending, CFPB complains, interpersonal trust and Madoff victims

 $Y_{i,t} = \beta_1 \text{CFPB complaints}_{i,t} + \beta_2 \text{Interpersonal trust}_i + \beta_3 \text{Madoff victims}_i + \gamma X_{i,t} + (B_t) + \epsilon_{i,t}.$

The dependent variable is the originated P2P debt amount in state i in month t normalized by total debt. *CFPB* is the number of consumer complaints per branch in state i in month t. *Interpersonal trust* is the measure of social trust from the General Social Survey (GSS) during 1973-2006. Madoff victims is the number (in thousands) of investors who suffered losses in the scandal at the state level. Variables *DTI*, *Interest rate*, *Income*, and *Employment* are the average measures for borrowers in state i in month t. We also control for state level variables population density $(1000/km^2)$, the logarithm of GDP, the logarithm of population and state unemployment rate. Standard errors, in parentheses, are corrected for clustering of observations by year-month. *, **, and *** indicate statistical significance at the 10%, 5%, and 1% level.

Earlier Working Papers:

For a complete list of Working Papers published by Sveriges Riksbank, see www.riksbank.se

Estimation of an Adaptive Stock Market Model with Heterogeneous Agents by Henrik Amilon	2005:177
Some Further Evidence on Interest-Rate Smoothing: The Role of Measurement Errors in the Output Gap by Mikael Apel and Per Jansson	2005:178
Bayesian Estimation of an Open Economy DSGE Model with Incomplete Pass-Through by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:179
Are Constant Interest Rate Forecasts Modest Interventions? Evidence from an Estimated Open Economy DSGE Model of the Euro Area by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:180
Inference in Vector Autoregressive Models with an Informative Prior on the Steady State by Mattias Villani	2005:181
Bank Mergers, Competition and Liquidity by Elena Carletti, Philipp Hartmann and Giancarlo Spagnolo	2005:182
Testing Near-Rationality using Detailed Survey Data by Michael F. Bryan and Stefan Palmqvist	2005:183
Exploring Interactions between Real Activity and the Financial Stance by Tor Jacobson, Jesper Lindé and Kasper Roszbach	2005:184
Two-Sided Network Effects, Bank Interchange Fees, and the Allocation of Fixed Costs by Mats A. Bergman	2005:185
Trade Deficits in the Baltic States: How Long Will the Party Last? by Rudolfs Bems and Kristian Jönsson	2005:186
Real Exchange Rate and Consumption Fluctuations follwing Trade Liberalization by Kristian Jönsson	2005:187
Modern Forecasting Models in Action: Improving Macroeconomic Analyses at Central Banks by Malin Adolfson, Michael K. Andersson, Jesper Lindé, Mattias Villani and Anders Vredin	2005:188
Bayesian Inference of General Linear Restrictions on the Cointegration Space by Mattias Villani	2005:189
Forecasting Performance of an Open Economy Dynamic Stochastic General Equilibrium Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:190
Forecast Combination and Model Averaging using Predictive Measures by Jana Eklund and Sune Karlsson	2005:191
Swedish Intervention and the Krona Float, 1993-2002 by Owen F. Humpage and Javiera Ragnartz	2006:192
A Simultaneous Model of the Swedish Krona, the US Dollar and the Euro by Hans Lindblad and Peter Sellin	2006:193
Testing Theories of Job Creation: Does Supply Create Its Own Demand? by Mikael Carlsson, Stefan Eriksson and Nils Gottfries	2006:194
Down or Out: Assessing The Welfare Costs of Household Investment Mistakes by Laurent E. Calvet, John Y. Campbell and Paolo Sodini	2006:195
Efficient Bayesian Inference for Multiple Change-Point and Mixture Innovation Models by Paolo Giordani and Robert Kohn	2006:196
Derivation and Estimation of a New Keynesian Phillips Curve in a Small Open Economy by Karolina Holmberg	2006:197
Technology Shocks and the Labour-Input Response: Evidence from Firm-Level Data by Mikael Carlsson and Jon Smedsaas	2006:198
Monetary Policy and Staggered Wage Bargaining when Prices are Sticky by Mikael Carlsson and Andreas Westermark	2006:199
The Swedish External Position and the Krona by Philip R. Lane	2006:200

Price Setting Transactions and the Role of Denominating Currency in FX Markets by Richard Friberg and Fredrik Wilander	2007:201
The geography of asset holdings: Evidence from Sweden by Nicolas Coeurdacier and Philippe Martin	2007:202
Evaluating An Estimated New Keynesian Small Open Economy Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2007:203
The Use of Cash and the Size of the Shadow Economy in Sweden by Gabriela Guibourg and Björn Segendorf	2007:204
Bank supervision Russian style: Evidence of conflicts between micro- and macro-prudential concerns by Sophie Claeys and Koen Schoors	2007:205
Optimal Monetary Policy under Downward Nominal Wage Rigidity by Mikael Carlsson and Andreas Westermark	2007:206
Financial Structure, Managerial Compensation and Monitoring by Vittoria Cerasi and Sonja Daltung	2007:207
Financial Frictions, Investment and Tobin's q by Guido Lorenzoni and Karl Walentin	2007:208
Sticky Information vs Sticky Prices: A Horse Race in a DSGE Framework by Mathias Trabandt	2007:209
Acquisition versus greenfield: The impact of the mode of foreign bank entry on information and bank lending rates by Sophie Claeys and Christa Hainz	2007:210
Nonparametric Regression Density Estimation Using Smoothly Varying Normal Mixtures by Mattias Villani, Robert Kohn and Paolo Giordani	2007:211
The Costs of Paying – Private and Social Costs of Cash and Card by Mats Bergman, Gabriella Guibourg and Björn Segendorf	2007:212
Using a New Open Economy Macroeconomics model to make real nominal exchange rate forecasts by Peter Sellin	2007:213
Introducing Financial Frictions and Unemployment into a Small Open Economy Model by Lawrence J. Christiano, Mathias Trabandt and Karl Walentin	2007:214
Earnings Inequality and the Equity Premium by Karl Walentin	2007:215
Bayesian forecast combination for VAR models by Michael K. Andersson and Sune Karlsson	2007:216
Do Central Banks React to House Prices? by Daria Finocchiaro and Virginia Queijo von Heideken	2007:217
The Riksbank's Forecasting Performance by Michael K. Andersson, Gustav Karlsson and Josef Svensson	2007:218
Macroeconomic Impact on Expected Default Freqency by Per Åsberg and Hovick Shahnazarian	2008:219
Monetary Policy Regimes and the Volatility of Long-Term Interest Rates by Virginia Queijo von Heideken	2008:220
Governing the Governors: A Clinical Study of Central Banks by Lars Frisell, Kasper Roszbach and Giancarlo Spagnolo	2008:221
The Monetary Policy Decision-Making Process and the Term Structure of Interest Rates by Hans Dillén	2008:222
How Important are Financial Frictions in the US and the Euro Area by Virginia Queijo von Heideken	2008:223
Block Kalman filtering for large-scale DSGE models by Ingvar Strid and Karl Walentin	2008:224
Optimal Monetary Policy in an Operational Medium-Sized DSGE Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Lars E. O. Svensson	2008:225
Firm Default and Aggregate Fluctuations by Tor Jacobson, Rikard Kindell, Jesper Lindé and Kasper Roszbach	2008:226
Re-Evaluating Swedish Membership in EMU: Evidence from an Estimated Model by Ulf Söderström	2008:227

The Effect of Cash Flow on Investment: An Empirical Test of the Balance Sheet Channel by Ola Melander	2009:228
Expectation Driven Business Cycles with Limited Enforcement by Karl Walentin	2009:229
Effects of Organizational Change on Firm Productivity by Christina Håkanson	2009:230
Evaluating Microfoundations for Aggregate Price Rigidities: Evidence from Matched Firm-Level Data on Product Prices and Unit Labor Cost by Mikael Carlsson and Oskar Nordström Skans	2009:231
Monetary Policy Trade-Offs in an Estimated Open-Economy DSGE Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Lars E. O. Svensson	2009:232
Flexible Modeling of Conditional Distributions Using Smooth Mixtures of Asymmetric Student T Densities by Feng Li, Mattias Villani and Robert Kohn	2009:233
Forecasting Macroeconomic Time Series with Locally Adaptive Signal Extraction by Paolo Giordani and Mattias Villani	2009:234
Evaluating Monetary Policy by Lars E. O. Svensson	2009:235
Risk Premiums and Macroeconomic Dynamics in a Heterogeneous Agent Model by Ferre De Graeve, Maarten Dossche, Marina Emiris, Henri Sneessens and Raf Wouters	2010:236
Picking the Brains of MPC Members by Mikael Apel, Carl Andreas Claussen and Petra Lennartsdotter	2010:237
Involuntary Unemployment and the Business Cycle by Lawrence J. Christiano, Mathias Trabandt and Karl Walentin	2010:238
Housing collateral and the monetary transmission mechanism by Karl Walentin and Peter Sellin	2010:239
The Discursive Dilemma in Monetary Policy by Carl Andreas Claussen and Øistein Røisland	2010:240
Monetary Regime Change and Business Cycles by Vasco Cúrdia and Daria Finocchiaro	2010:241
Bayesian Inference in Structural Second-Price common Value Auctions by Bertil Wegmann and Mattias Villani	2010:242
Equilibrium asset prices and the wealth distribution with inattentive consumers by Daria Finocchiaro	2010:243
Identifying VARs through Heterogeneity: An Application to Bank Runs by Ferre De Graeve and Alexei Karas	2010:244
Modeling Conditional Densities Using Finite Smooth Mixtures by Feng Li, Mattias Villani and Robert Kohn	2010:245
The Output Gap, the Labor Wedge, and the Dynamic Behavior of Hours by Luca Sala, Ulf Söderström and Antonella Trigari	2010:246
Density-Conditional Forecasts in Dynamic Multivariate Models by Michael K. Andersson, Stefan Palmqvist and Daniel F. Waggoner	2010:247
Anticipated Alternative Policy-Rate Paths in Policy Simulations by Stefan Laséen and Lars E. O. Svensson	2010:248
MOSES: Model of Swedish Economic Studies by Gunnar Bårdsen, Ard den Reijer, Patrik Jonasson and Ragnar Nymoen	2011:249
The Effects of Endogenuos Firm Exit on Business Cycle Dynamics and Optimal Fiscal Policy by Lauri Vilmi	2011:250
Parameter Identification in a Estimated New Keynesian Open Economy Model by Malin Adolfson and Jesper Lindé	2011:251
Up for count? Central bank words and financial stress by Marianna Blix Grimaldi	2011:252
Wage Adjustment and Productivity Shocks by Mikael Carlsson, Julián Messina and Oskar Nordström Skans	2011:253

Stylized (Arte) Facts on Sectoral Inflation by Ferre De Graeve and Karl Walentin	2011:254
Hedging Labor Income Risk by Sebastien Betermier, Thomas Jansson, Christine A. Parlour and Johan Walden	2011:255
Taking the Twists into Account: Predicting Firm Bankruptcy Risk with Splines of Financial Ratios by Paolo Giordani, Tor Jacobson, Erik von Schedvin and Mattias Villani	2011:256
Collateralization, Bank Loan Rates and Monitoring: Evidence from a Natural Experiment by Geraldo Cerqueiro, Steven Ongena and Kasper Roszbach	2012:257
On the Non-Exclusivity of Loan Contracts: An Empirical Investigation by Hans Degryse, Vasso Ioannidou and Erik von Schedvin	2012:258
Labor-Market Frictions and Optimal Inflation by Mikael Carlsson and Andreas Westermark	2012:259
Output Gaps and Robust Monetary Policy Rules by Roberto M. Billi	2012:260
The Information Content of Central Bank Minutes by Mikael Apel and Marianna Blix Grimaldi	2012:261
The Cost of Consumer Payments in Sweden	2012:262
by Björn Segendorf and Thomas Jansson	
Trade Credit and the Propagation of Corporate Failure: An Empirical Analysis	2012:263
by Tor Jacobson and Erik von Schedvin	
Structural and Cyclical Forces in the Labor Market During the Great Recession: Cross-Country Evidence	2012:264
by Luca Sala, Ulf Söderström and AntonellaTrigari	
Pension Wealth and Household Savings in Europe: Evidence from SHARELIFE	2013:265
by Rob Alessie, Viola Angelini and Peter van Santen	
Long-Term Relationship Bargaining	2013:266
by Andreas Westermark	
Using Financial Markets To Estimate the Macro Effects of Monetary Policy: An Impact-Identified FAVAR*	2013:267
by Stefan Pitschner	
DYNAMIC MIXTURE-OF-EXPERTS MODELS FOR LONGITUDINAL AND DISCRETE-TIME SURVIVAL DATA	2013:268
by Matias Quiroz and Mattias Villani	
Conditional euro area sovereign default risk	2013:269
by André Lucas, Bernd Schwaab and Xin Zhang	
Nominal GDP Targeting and the Zero Lower Bound: Should We Abandon Inflation Targeting?*	2013:270
by Roberto M. Billi	
Un-truncating VARs*	2013:271
by Ferre De Graeve and Andreas Westermark	
Housing Choices and Labor Income Risk	2013:272
by Thomas Jansson	
Identifying Fiscal Inflation*	2013:273
by Ferre De Graeve and Virginia Queijo von Heideken	0040.074
On the Redistributive Effects of Inflation: an International Perspective"	2013:274
by Paola Boel	0040.075
Business Cycle Implications of Mongage Spreads	2013:275
by Kall Walefull	2012:276
economic models by Isajah Hull	2013.270
A detrimental feedback loop: deleveraging and adverse selection	2013-277
by Christoph Bartsch	2013.211
Distortionary Fiscal Policy and Monetary Policy Goals	2013.278
by Klaus Adam and Roberto M Billi	2010.270
Predicting the Spread of Financial Innovations: An Enidemiological Approach	2013.279
by Isaiah Hull	
Firm-Level Evidence of Shifts in the Supply of Credit	2013:280
· · · · · · · · · · · · · · · · · · ·	

Lines of Credit and Investment: Firm-Level Evidence of Real Effects of the Financial Crisis by Karolina Holmberg	2013:281
A wake-up call: information contagion and strategic uncertainty by Toni Ahnert and Christoph Bertsch	2013:282
Debt Dynamics and Monetary Policy: A Note	2013:283
by Stefan Laséen and Ingvar Strid	
Optimal taxation with home production	2014:284
by Conny Olovsson	
Incompatible European Partners? Cultural Predispositions and Household Financial Behavior	2014:285
by Michael Haliassos. Thomas Jansson and Yigitcan Karabulut	
How Subprime Borrowers and Mortgage Brokers Shared the Piecial Behavior	2014:286
by Antie Berndt, Burton Hollifield and Patrik Sandås	
The Macro-Financial Implications of House Price-Indexed Mortgage Contracts	2014:287
by Isaiah Hull	
Does Trading Anonymously Enhance Liquidity?	2014:288
by Patrick J. Dennis and Patrik Sandås	
Systematic bailout guarantees and tacit coordination	2014:289
by Christoph Bertsch. Claudio Calcagno and Mark Le Quement	
Selection Effects in Producer-Price Setting	2014.290
by Mikael Carlsson	2011.200
Dynamic Demand Adjustment and Exchange Rate Volatility	2014:291
by Vesna Corbo	
Forward Guidance and Long Term Interest Rates: Inspecting the Mechanism	2014.292
by Ferre De Graeve. Pelin Ilbas & Raf Wouters	2011202
Firm-I evel Shocks and Labor Adjustments	2014.293
by Mikael Carlsson, Julián Messina and Oskar Nordström Skans	2011200
A wake-up call theory of contagion	2015:294
by Toni Ahnert and Christoph Bertsch	2010.201
Risks in macroeconomic fundamentals and excess bond returns predictability	2015:295
by Rafael B. De Rezende	2010.200
The Importance of Reallocation for Productivity Growth: Evidence from European and US Banking	2015:296
by Jaan W.B. Bos and Peter C. van Santen	2010.200
SPEEDING LIP MCMC BY EFFICIENT DATA SUBSAMPLING	2015:297
by Matias Quiroz Mattias Villani and Robert Kohn	2010.201
Amortization Requirements and Household Indeptedness: An Application to Swedish-Style Mortgages	2015:208
by Isaiah Hull	2013.290
Fuel for Economic Growth?	2015:299
by Johan Gars and Conny Olovsson	2010.200
Searching for Information	2015:300
by Jungsuk Han and Francesco Sangiorgi	2010.000
What Broke First? Characterizing Sources of Structural Change Prior to the Great Recession	2015:301
by Isaiah Hull	2010.001
Price Level Targeting and Risk Management	2015:302
by Roberto Billi	2010.002
Central bank policy paths and market forward rates: A simple model	2015:303
by Ferre De Graeve and Jens Iversen	2010.000
Jump-Starting the Furn Area Recovery: Would a Rise in Core Fiscal Spending Halp the Perinhery?	2015:304
by Olivier Blanchard Christonher I Erreg and Jesper Lindé	2010.004
Bringing Financial Stability into Monatary Policy*	2015.205
by Fric M Leener and James M Nason	2013.305
SCALABLE MCMC FOR LARGE DATA PROBLEMS USING DATA SUBSAMPLING AND	2015:306
by MATIAS QUIROZ. MATTIAS VILLANI AND ROBERT KOHN	

SPEEDING UP MCMC BY DELAYED ACCEPTANCE AND DATA SUBSAMPLING by MATIAS QUIROZ	2015:307
Modeling financial sector joint tail risk in the euro area by André Lucas, Bernd Schwaab and Xin Zhang	2015:308
Score Driven Exponentially Weighted Moving Averages and Value-at-Risk Forecasting	2015:309
by André Lucas and Xin Zhang	
On the Theoretical Efficacy of Quantitative Easing at the Zero Lower Bound	2015:310
by Paola Boel and Christopher J. Waller	
Optimal Inflation with Corporate Taxation and Financial Constraints	2015:311
by Daria Finocchiaro, Giovanni Lombardo, Caterina Mendicino and Philippe Weil	
Fire Sale Bank Recapitalizations	2015:312
by Christoph Bertsch and Mike Mariathasan	
Since you're so rich, you must be really smart: Talent and the Finance Wage Premium	2015:313
by Michael Böhm, Daniel Metzger and Per Strömberg	
Debt, equity and the equity price puzzle	2015:314
by Daria Finocchiaro and Caterina Mendicino	
Trade Credit: Contract-Level Evidence Contradicts Current Theories	2016:315
by Tore Ellingsen, Tor Jacobson and Erik von Schedvin	
Double Liability in a Branch Banking System: Historical Evidence from Canada	2016:316
by Anna Grodecka and Antonis Kotidis	
Subprime Borrowers, Securitization and the Transmission of Business Cycles	2016:317
by Anna Grodecka	
Real-Time Forecasting for Monetary Policy Analysis: The Case of Sveriges Riksbank	2016:318
by Jens Iversen, Stefan Laséen, Henrik Lundvall and Ulf Söderström	
Fed Liftoff and Subprime Loan Interest Rates: Evidence from the Peer-to-Peer Lending	2016:319
by Christoph Bertsch, Isaiah Hull and Xin Zhang	
Curbing Shocks to Corporate Liquidity: The Role of Trade Credit	2016:320
by Niklas Amberg, Tor Jacobson, Erik von Schedvin and Robert Townsend	
Firms' Strategic Choice of Loan Delinquencies	2016:321
by Paola Morales-Acevedo	
Fiscal Consolidation Under Imperfect Credibility	2016:322
by Matthieu Lemoine and Jesper Lindé	
Challenges for Central Banks' Macro Models	2016:323
by Jesper Lindé, Frank Smets and Rafael Wouters	
The interest rate effects of government bond purchases away from the lower bound	2016:324
by Rafael B. De Rezende	
COVENANT-LIGHT CONTRACTS AND CREDITOR COORDINATION	2016:325
by Bo Becker and Victoria Ivashina	
Endogenous Separations, Wage Rigidities and Employment Volatility	2016:326
by Mikael Carlsson and Andreas Westermark	
Renovatio Monetae: Gesell Taxes in Practice	2016:327
by Roger Svensson and Andreas Westermark	
Adjusting for Information Content when Comparing Forecast Performance	2016:328
by Michael K. Andersson, Ted Aranki and André Reslow	
Economic Scarcity and Consumers' Credit Choice	2016:329
by Marieke Bos, Chloé Le Coq and Peter van Santen	
Uncertain pension income and household saving	2016:330
by Peter van Santen	
Money, Credit and Banking and the Cost of Financial Activity	2016:331
by Paola Boel and Gabriele Camera	
Oil prices in a real-business-cycle model with precautionary demand for oil	2016:332
by Conny Olovsson	
Financial Literacy Externalities	2016:333
by Michael Haliasso, Thomas Jansson and Yigitcan Karabulut	

The timing of uncertainty shocks in a small open economy	2016:334
by Hanna Armelius, Isaiah Hull and Hanna Stenbacka Köhler	
Quantitative easing and the price-liquidity trade-off	2017:335
by Marien Ferdinandusse, Maximilian Freier and Annukka Ristiniemi	
What Broker Charges Reveal about Mortgage Credit Risk	2017:336
by Antje Berndt, Burton Hollifield and Patrik Sandåsi	
Asymmetric Macro-Financial Spillovers	2017:337
by Kristina Bluwstein	
Latency Arbitrage When Markets Become Faster	2017:338
by Burton Hollifield, Patrik Sandås and Andrew Todd	
How big is the toolbox of a central banker? Managing expectations with policy-rate forecasts:	2017:339
Evidence from Sweden	
by Magnus Ahl	
International business cycles: quantifying the effects of a world market for oil	2017:340
by Johan Gars and Conny Olovsson I	
Systemic Risk: A New Trade-Off for Monetary Policy?	2017:341
by Stefan Laséen, Andrea Pescatori and Jarkko Turunen	
Household Debt and Monetary Policy: Revealing the Cash-Flow Channel	2017:342
by Martin Flodén, Matilda Kilström, Jósef Sigurdsson and Roine Vestman	
House Prices, Home Equity, and Personal Debt Composition	2017:343
by Jieying Li and Xin Zhang	
Identification and Estimation issues in Exponential Smooth Transition Autoregressive Models	2017:344
by Daniel Buncic	
Domestic and External Sovereign Debt	2017:345
by Paola Di Casola and Spyridon Sichlimiris	



Sveriges Riksbank Visiting address: Brunkebergs torg 11 Mail address: se-103 37 Stockholm

Website: www.riksbank.se Telephone: +46 8 787 00 00, Fax: +46 8 21 05 31 E-mail: registratorn@riksbank.se